INFORMATION TO USERS

This manuscript has been reproduced from the microfilm master. UMI

films the text directly from the original or copy submitted. Thus, some

thesis and dissertation copies are in typewriter face, while others may be

from any type of computer printer.

The quality of this reproduction is dependent upon the quality of the

copy submitted. Broken or indistinct print, colored or poor quality

illustrations and photographs, print bleedthrough, substandard margins,

and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send UMI a complete

manuscript and there are missing pages, these will be noted. Also, if

unauthorized copyright material had to be removed, a note will indicate

the deletion.

Oversize materials (e.g., maps, drawings, charts) are reproduced by

sectioning the original, beginning at the upper left-hand corner and

continuing from left to right in equal sections with small overlaps. Each

original is also photographed in one exposure and is included in reduced

form at the back of the book.

Photographs included in the original manuscript have been reproduced

xerographically in this copy. Higher quality 6" x 9" black and white

photographic prints are available for any photographs or illustrations

appearing in this copy for an additional charge. Contact UMI directly to

order.

IIMI

A Bell & Howell Information Company 300 North Zeeb Road, Ann Arbor MI 48106-1346 USA

313/761-4700 800/521-0600



FIRMS' USE OF DERIVATIVE SECURITIES: AN EXAMINATION OF THE FIRM-SPECIFIC IMPACT OF DERIVATIVE STRATEGIES ON EARNINGS PERSISTENCE AND EARNINGS RESPONSE COEFFICIENTS

Jennifer Ann Reynolds-Moehrle

submitted to the Faculty of the University Graduate School
in Partial Fulfillment of the Requirements
for the Degree
Doctor of Philosophy
in the School of Business
Indiana University

August, 1997

UMI Number: 9805431

Copyright 1997 by Reynolds-Moehrle, Jennifer Ann

All rights reserved.

UMI Microform 9805431 Copyright 1997, by UMI Company. All rights reserved.

This microform edition is protected against unauthorized copying under Title 17, United States Code.

300 North Zeeb Road Ann Arbor, MI 48103

ACCEPTANCE

Accepted by the Graduate Faculty, Indiana University, in partial fulfillment of the requirements of the Degree of Doctor of Philosophy.

Gerald L. Salamon, Chairman

Walter G. Blacconiere

Craig W. Holden

Mike G Tille

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Doctoral Committee

July 25, 1997

c Jennifer Ann Reynolds-Moehrle 1997 All Rights Reserved

ACKNOWLEDGMENTS

I would like to gratefully acknowledge the encouragement and help of those who made this endeavor possible.

I would like to thank my chairman, Gerald L. Salamon, for the guidance he has provided throughout this process and throughout the doctoral program. His friendship, encouragement, and advice have been invaluable. I would also like to thank Walter G. Blacconiere for his support and encouragement. His involvement made the doctoral program and the dissertation process enjoyable and I appreciate his friendship. I greatly appreciate the insights provided by Craig W. Holden. His enthusiasm aided in my interest in the finance area and he helped me integrate my knowledge of accounting and finance. I would like to express my gratitude to Mikel G. Tiller for his advice and genuine concern during the doctoral program and the dissertation. I would also like to thank Thomas J. Linsmeier for his insights into the area of derivative securities. His knowledge helped me in the development of the ideas presented in this dissertation.

I would like to thank the Ernst and Young Foundation for financial support during the dissertation process. I would also like to thank the University of Indiana School of Business for their financial support during the doctoral program.

I would like to thank my family for their unfailing encouragement and support.

Without their love and guidance, I could not have accomplished what I have. I thank

Mom and Dad for instilling in me the belief that I can do anything I set out to do. I

would like to thank my brother, Doug, for his humor and countless hours of driving

between Bloomington and Owensboro to help me whenever I asked. I thank my brother,

Bryan, for motivating me to prove that his expectations for me are warranted. And I would like to thank my sister, Suzanne, for her friendship and support throughout this doctoral program. I am very lucky to have her.

I would like to thank my husband, Steve, for adding so much to my life and for making this process worthwhile. I thank Allison for keeping a smile on her face throughout the last year while her mom and dad have been so busy. And finally, I would especially like to thank one little girl, Elizabeth, for inspiring her mom to take this chance and for helping to keep life in perspective during the hardest parts of this program.

ABSTRACT

Use of derivative securities has increased exponentially over the last 10 years. The purpose of this dissertation is to predict the impact of a firm's interest rate derivative strategy on the valuation relevance of its earnings and to empirically test the predictions using a sample of non-financial firms. The model describes the impact of an interest rate derivative strategy on the persistence of earnings and on the earnings response coefficient. The empirical tests examine changes in persistence and earnings response coefficients for a sample of non-financial firms that have disclosed the use of interest rate swap contracts.

The model describes the earnings process as a combination of two components: one component that is related to interest rates and one component that is independent of interest rates. The model shows that total persistence of earnings is a weighted average of the persistence of each of the components and that total persistence increases when the component that is related to interest rate risk is decreased. When an effective hedging strategy is defined as one that reduces the exposure to interest rate risk, the model predicts that implementation of an effective hedging strategy leads to an increase in earnings persistence.

Earnings persistence has been examined in prior accounting research and has been found to be directly related to earnings response coefficients. Therefore, the second prediction made in this dissertation is that implementation of an effective hedging strategy leads to an increase in the earnings response coefficient.

This dissertation includes empirical tests of the predictions regarding earnings persistence and earnings response coefficients. The tests use regression analysis to examine the change in the relation between analysts' revisions of future earnings and current unexpected earnings following disclosure of an interest rate swap strategy (the persistence test). Also examined is the change in the relation between abnormal returns and unexpected earnings following disclosure of an interest rate swap (the earnings response coefficient test). The empirical results support the theoretical predictions.

TABLE OF CONTENTS

ABSTRACT

LIST OF TABLES

LIST OF EXHIBITS

I.	INTRODUCTION	1
II.	THE VALUE RELEVANCE OF HEDGING AT THE FIRM LEVEL	4
	2.1 Introduction	4
	2.2 The possible value relevance of firm level hedging	4
	2.2.1 Asymmetric information and agency cost theories of firm level hedging	5
	2.2.2 Manager compensation contracts and firm level hedging	6
	2.2.3 Tax savings as a motivation for firm level hedging	8
	2.2.4 Financial distress costs as a motivation for firm level	
	hedging	8
	2.2.5 Investment protection as a motivation for firm level	
	hedging	9
	2.3 Summary	12
III.	INTEREST RATE SWAPS AS PART OF A HEDGING STRATEGY	13
	3.1 Introduction	13
	3.2 Issues surrounding derivative strategies	13
	3.3 Mechanics of Interest Rate Swaps and Potential Cash Flow Impacts	17
	3.3.1 Example of Interest rate swap mechanics	17
	3.3.2 Definition of effective and ineffective cash flow hedging	19
	3.4 Accounting treatment for interest rate swaps	20
	3.4.1 Current accounting treatment	20
	3.4.2 Example of financial statement recognition and disclosure	23
	3.4.3 Proposed changes to accounting treatment of interest rate	
	swaps	28
	3.5 Summary	30
T	THE IMPACT OF A DERIVATIVE STRATEGY ON THE EARNINGS	
IV.	PROCESS	31
	4.1 Introduction	31
	· · · · · · · · · · · · · · · · · · ·	31
	4.2 The valuation relevance of the earnings stream	32
	4.3 A model of persistence in the quarterly earnings process	39
	4.4 Interest rate derivative strategies and persistence	
	4.5 Interest rate derivative strategies and earnings response coefficients	41 44
	4.6 Summary	44

V.	SAMPLE SELECTION, CATEGORIZATION OF FIRMS, AND	
	DESCRIPTIVE STATISTICS	45
	5.1 Introduction	45
	5.2 Sample Selection	45
	5.3 Categorization of firms	48
	5.3.1 Model of interest rate sensitivity	48
	5.3.2 Pre-period interest sensitivity	48
	5.3.2 Post-period interest sensitivity	49
	5.3.4 Decision rule for grouping firms	50
	5.3.5 Disclosure	52
	5.3.6 Final grouping	52
	5.4 Statistics for categories	53
	5.4.1 Univariate statistics across all time	55
	5.4.2 Test of means between treatment and control	55
	5.4.3 Tests of differendes between pre- and post-swapping	
	time periods	56
	5.4.4 Statistics for eliminated firms	60
	5.4.5 Industry representation	60
	5.4.6 Economic significance of interest rate swaps	60
	5.5 Summary	65
VI.	EARNINGS PERSISTENCE: EMPIRICAL MODEL AND REGRESSIO	N
	RESULTS	66
	6.1 Introduction	66
	6.2 Earnings persistence model	66
	6.3 Test of persistence response	69
	6.3.1 Regresssion results for Model 1	69
	6.3.2 Regression results for Model 1 with firm-specific and time-	
	specific indicators	72
	6.4 Summary	75
VII.	EARNINGS RESPONSE COEFFICIENT EMPIRICAL MODEL AND	
	REGRESSION RESULTS	76
	7.1 Introduction	76
	7.2 The earnings response coefficient model	76
	7.3.Test of ERC response	78
	7.3.1 Regression results for Model 2	78
	7.3.2 Regression results for Model 2 with firm-specific and time-	
	specific indicators	82
	7.4 Summary	83

VIII.	CONCLUSION AND SUGGESTIONS FOR FUTURE RESEARCH	86
	8.1 Interpretation of results	86
	8.2 Limitations and suggestions for future research	87
APPE	NDIX A	89
RFFF	RENCES	93

LIST OF EXHIBITS

Exhibit 3.1 Lizco example of interest rate swap mechanics	18
Exhibit 3.2 Effective and Ineffective cash flow hedging	19
Exhibit 3.3 Lizco interest rates	24
Exhibit 3.4 Accounting entries related to interest rate swaps	25
Exhibit 3.5 Financial Statement Impact 12/31/x1	26
Exhibit 3.6 Financial Statement Impact Next three years	26
Exhibit 3.7 Lizco Disclosure	27
Exhibit 3.8 Financial Statement Impact Proposed accounting treatment	28
Exhibit 4.1 Values of Persistence	39
Exhibit 4.2 Predictions for persistence and ERCs	44
Exhibit 5.1 Disclosure example	53
Exhibit 6.1 Coefficients in Model 1	69
Exhibit 7.1 Coefficients in Model 2	78

84

variables for each firm and time period

CHAPTER I

INTRODUCTION

Use of derivative securities has increased exponentially over the last 10 years. On July 17, 1997, the Wall Street Journal (page A2) quoted Arthur C. Levitt, chairman of the Securities and Exchange Commission as saying, "We are in a situation today in which the notional amount of derivatives outstanding has reached some \$70 trillion". The nature of derivatives contracts combined with highly individualized financial strategies across firms makes financial evaluation extremely complex. Currently, both the Securities and Exchange Commission (SEC) and the Financial Accounting Standards Board (FASB) are working towards providing appropriate accounting guidance for this dynamic class of financial instruments. This dissertation examines the most widely used derivative security, the interest rate swap, and investigates the effect of the firm's interest rate swap strategy on both the persistence of earnings and the earnings-return relation.

While most of the prior research in this area has investigated usage of derivative securities by financial institutions (e.g., Schrand (1996), Riffe (1995),

¹Notional value is a contract's face value, but is not exchanged in cash in derivative transactions. Although these values are not "cash flows", the cash flows implied by the notional values is still very significant. For example, if all outstanding contracts were interest rate swaps and the average interest rate was 10 percent, then the gross cash flows associated with the notional value of \$70 trillion would be approximately \$7 trillion per year. In addition, the reported estimate of notional value does not include the value associated with contracts that are traded on exchanges (e.g., exchange traded futures and options).

²In response to <u>The Wharton School/Chase survey of Derivatives Usage Among U.S. Non-Financial Firms</u> (1995), firms responded that 67.2% used hedging strategies to manage cash flow exposure (survey question 5 response). In addition, question 3 of the survey asked firms which derivatives they used to manage interest rate risk and 62.8% of firms responded that interest rate swaps were used to manage interest rate risk.

Venkatachalam(1996)), this dissertation focuses on interest rate swaps and their effect on the persistence of earnings for a group of non-financial firms. Interest rate swaps alter the earnings pattern of the firm and thus, have implications for earnings persistence and earnings response coefficients (ERCs). In this study, a model is developed that describes the impact of interest rate swaps on earnings persistence and on ERCs.

The empirical predictions generated by the model are used to evaluate the positions taken by a group of non-financial firms. Evidence produced by this research is potentially relevant to assessing whether end users enter into such agreements for cash flow hedging purposes or for purposes inconsistent with cash flow hedging. The findings suggest that interest rate derivative strategies taken by firms are valuation relevant. Interest rate swaps that were part of an "effective" cash flow hedging strategy were found to increase earnings persistence and to increase ERCs while interest rate swaps that were not part of an "effective" cash flow hedging strategy were found to have no impact on either the earnings persistence or ERCs.

The evidence provided by this dissertation also adds additional support to the earlier work of Hentschel and Kothari (1995) who concluded that firms were not gambling on average with their derivative strategies. Using directional predictions for the impact on persistence and ERCs of effectively hedged cash flows, the results of this dissertation support a hedging strategy, thus strengthening the position that on average, non-financial firms are not speculating on cash flows with their interest rate swaps.

The dissertation proceeds as follows. Chapter II presents a review of the finance literature regarding the value relevance of a firm level hedging strategy. The purpose of

this literature review is to provide support for the underlying assumption made in this dissertation, that the firm value is increased by an effective hedging strategy. A discussion of the issues surrounding the reporting of derivative strategies and a description of potential cash flow impacts from an interest rate swap strategy are presented in Chapter III. In addition, an example is used to demonstrate the mechanics of an interest rate swap strategy and the accounting treatment under relevant reporting requirements. The purpose of Chapter III is to demonstrate the direct relation between the cash flow effects of an interest rate swap strategy and the reported financial position impact of the strategy. In Chapter IV, the possible effects of interest rate swaps on the earnings process are examined and specific predictions about how swaps will impact persistence and the earnings response coefficient are developed. These predictions are the bases for the empirical tests performed in Chapters VI and VII. The purpose of Chapter V is to provide a description of the sample selection process and to present descriptive statistics for the firms included in the empirical tests. An integral part of Chapter V is the description of the process used to categorize firms as effective or ineffective cash flow hedgers. Chapter VI presents the empirical model used to test the predictions regarding changes in persistence following implementation of an interest rate swap strategy and discusses the results of running the model using the sample described in Chapter V. Chapter VII presents the empirical model used to test the predictions regarding the earnings response coefficient response to an interest rate swap strategy and discusses the results of running the model. Chapter VIII concludes and discusses future research opportunities.

CHAPTER II

THE VALUE RELEVANCE OF HEDGING AT THE FIRM LEVEL

2.1 Introduction

This dissertation addresses the valuation relevance of a particular type of hedging strategy. However, the valuation relevance only becomes an issue after the firm has made the decision to hedge. Many theories have emerged in the finance literature explaining possible motives for firms to hedge and several of these are presented in this chapter. The theoretical predictions and empirical examinations of this dissertation start from the underlying assumption that firm owners benefit from an effective cash flow hedging strategy. This chapter provides support for this underlying assumption.

2.2 The possible value relevance of firm level hedging

Traditional finance theory (Modigliani and Miller (1958)) is inconsistent with a strategy to hedge cash flows. Risk management is not needed at the firm level because investors can diversify their risks. This traditional theory is based on assumptions that lead to capital structure irrelevance (i.e. the financing of an investment does not add to or lessen the value of the firm, only the quality of the investment itself affects firm value). Primary among these assumptions is the perfection of capital markets.³ If, however, the market is allowed to have some imperfect qualities, a role emerges for firm level hedging.

Several theories have been put forth to support a role for hedging (or for

³This proposition of capital structure irrelevance is presented in Modigliani and Miller (1958) as Proposition 1 and holds in the absence of any market imperfections (p. 268).

specifically using interest rate swaps) at the firm level and each depends upon certain market imperfections. These theories provide alternative motivations for hedging using asymmetric information scenarios, agency cost issues, progressive tax rate arguments, transactions costs issues and investment protection ideas. Each is discussed below along with a summary of the related empirical findings.

2.2.1 Asymmetric information and agency cost theories of firm level hedging

Within the asymmetric information scenarios and the agency cost theories, hedging allows firms to lower contracting costs while maintaining protection from interest rate risk. In the asymmetric information scenarios (Arak, Estrella, Goodman and Silver (1988), Titman (1992)), the firm's choice of debt contracts is used to signal private information about the firm's future credit premium. For example, the equilibrium reached in Titman's model has fixed rate long-term debt being used exclusively by firms with bad private information about their future credit quality. Firms with good private information (i.e., firms that expect to have better credit quality in the future) issue short-term debt. Hedging is used by these good private information firms to avoid the fluctuation in the short-term nominal interest rate.

One argument based on agency cost theory suggests that bondholders price protect against possible future underinvestment under a long-term fixed rate debt structure (Wall (1989)). This price protection decreases when shareholders have to go to the debt market often or when shareholders face variable rate debt (where the variable rate debt must include a variable credit quality premium). A combination of short-term debt or variable rate debt plus a hedge of the nominal rate changes acts as a substitute for

long-term fixed rate debt that allows for credit quality changes. Again, as in the asymmetric information story above, the hedge is valuable in this scenario because it allows for lower borrowing costs while maintaining protection from interest rate risk. Both theories predict that interest rate hedges should be used by firms with short-term debt.

The debt substitution theory has been examined empirically by Fenn, Post, and Sharpe (1996a) in their examination of debt maturity and the use of interest rate derivatives. The evidence indicates that the short-term debt/total debt is 10 to 15 percentage points higher for users of interest rate derivative users. This is interpreted as evidence supporting derivative use as increasing real cash flows (i.e., lowering borrowing costs), to the extent that short-term debt carries a lower credit risk premium than long-term debt. Fenn, Post, and Sharpe (1996a) note, however, that it is "difficult to distinguish empirically whether firms increase their use of short-term debt (leverage) because they use derivatives, or whether firms with greater short-term debt (leverage) are simply more likely to use derivatives." The authors further note that an examination of firms over time might provide a better test of the debt substitution hypothesis.

2.2.2 Manager compensation contracts and firm level hedging

A separate agency cost theory explanation for hedging focusses on managers' risk aversion. Smith and Stulz (1985) describe the case where hedging is valuable because it can remove fluctuations in performance-based compensation plans that are beyond the manager's control. The manager then does not have to be compensated for bearing these risks and the shareholders are better off (as long as the savings in compensation exceed

the costs of hedging). This theory, however, overlooks the potentially less costly alternative of hedging "on paper" only, an alternative that could accomplish the same goal as far as managers are concerned without increasing the number of actual hedging contracts used by firms.

Alternatively, one line of research offers that managers are motivated to hedge corporate earnings so that the capital markets can better judge the management's performance. Demarzo and Duffie (1991) use an information asymmetry model where managers have private information about the source and magnitude of the risks faced by the firm and may have private information about the hedging positions. The model develops an equilibrium where investors support managers' hedging because the decrease in the firms' expected profits is offset by the benefits of reduced noise in the investors' information sets regarding the variance of firm payoffs. In the model, the manager's incentive to hedge is negatively affected by the degree of disclosure required, thus explicitly including a role for accounting standards in shaping management's incentives to use derivatives.

Empirical evidence regarding the managerial risk aversion hypotheses has been mixed. As discussed by Fenn, Post, and Sharpe (1996b), conclusions are difficult to draw without knowing the managers' and shareholders' ex ante positions regarding managerial risk aversion. Geczy, Minton, and Schrand (1995) examined the relation between information asymmetry and derivative use and did not find support for the Demarzo and Duffie (1991) prediction that firms with higher information asymmetry would be more likely to hedge.

2.2.3 Tax savings as a motivation for firm level hedging

Within the previously described theories the value of the hedge comes primarily from its impact on future contracting costs, not from its direct impact on the risk being hedged. In contrast, the following theories propose that the reduction of variance achieved by the hedge adds value to the firm.

The progressive tax rate theory (Smith and Stulz (1985)) relies upon the intuition of taxes as a convex function of earnings. The convexity is due to two factors, as pointed out by Fenn, Post, and Sharpe (1996b), faced by corporations. One is the increasing statutory marginal tax rate (15% on the first \$100,000 and 34% on all income beyond) and the other is the existence of tax preference items (such as investment tax credits and loss carryforwards). Reducing the variability of earnings through a hedging instrument leads to lower expected taxes, or higher expected future cash flows and thus a higher firm value. As discussed by Fenn, Post and Sharpe (1996b), the firms that could realize the largest tax benefit from hedging are those whose income is especially volatile and/or those who are near a shift in the marginal tax rate.

Using data on firms' tax preference items and their pre-tax income, empirical studies have investigated this tax rate theory. Nance, Smith and Smithson 91993) found no statistically significant relation between derivative use and tax preference items. They concluded that confidence in the tax hypothesis is reduced based upon their results.

2.2.4 Financial distress costs as a motivation for firm level hedging

There are exogenous costs associated with financial distress that create a concave expected profit function. Financial distress costs include not only the relatively small

direct costs of bankruptcy but also the potentially material indirect costs of being close to bankruptcy. Hedging reduces the expected costs of financial distress by reducing the likelihood of cash shortages and consequently reduces the cost of debt financing (Smith and Stulz (1985)).

Empirically, the hypothesized positive relation between derivatives and leverage has been investigated and a positive relation has been found. Generally, however, the relation has been statistically and economically insignificant (Fenn, Post, and Sharpe (1996b)). Therefore, financial distress does not seem to be the key motivator for the growth in derivative use.

2.2.5 Investment protection as a motivation for firm level hedging

A final theorized value-relevant role for hedging comes from the perspective of investment protection. Froot, Scharfstein, and Stein (1993) develop a model of optimal hedging at the firm level based on the Myers and Majluf (1984) "pecking order" model.

The Myers and Majluf model assumes managers make financing and investment choices to maximize the wealth of passive, existing ("old") shareholders and assumes managers have information that shareholders do not have. The pecking order model suggests that old shareholders prefer to use internally generated funds for new projects when given a choice between internal funds or issuing new equity. This is because there is a cost to the old shareholders of issuing new equity. When new equity is issued to fund a project, the old shareholders give up part of the value of existing assets and gain only

⁴Passive in this context means a shareholder that does not rebalance his/her portfolio upon a new equity issuance by the firm.

part of the value of the new investment. If the value of the assets before any new investment exceeds the old shareholders' post-equity issuance share of the value of the assets plus the value of the new investment, the old shareholders are better off if no new equity is issued. However, this decision process results in some new projects being rejected only because internal funds are not available for an otherwise beneficial project. Myers and Majluf (1984) discuss "financial slack" as being cash, liquid assets, or unused borrowing power in addition to the existing productive assets and show that this "slack" has value because as the slack increases, the availability of internal funding increases, and the firm passes up fewer positive NPV projects without going to the equity market.⁵

Froot et. al. (1993) show that the value of hedging comes from the desire to protect the "slack". The optimal hedging strategy model developed provides for protection of internal funds (the "slack" in Myers and Majluf) when facing changing investment opportunity sets and changing financing opportunities. If internal funds are subject to external risks (such as changing interest rates), the cash flows associated with these funds will fluctuate. Hedging protects internal funds by removing the fluctuation. Relying on the assumption that profit is a concave function of internal wealth (because of the exogenous costs of financial distress), the protection of internal wealth increases the

⁵The pecking order model also is extended to rank debt financing between equity financing and internal funding. This theory appears to be empirically documented by the findings of an average decline in stock price after the announcement of new equity issues (Dann and Mikkelson (1984), Korwar (1982), Asquith and Mullins (1983)) and no apparent price reaction for the issuance of new debt securities (Dann and Mikkelson (1984)). In an experimental setting, Hopkins (1996) documents stock price predictions by analysts that also support the pecking order predictions.

shareholders' expectation of future profits and increases firm value.⁶ The intuition from the model is that shareholders will benefit from hedging at the firm level because firms will be more likely to accept positive net present value projects if the cash flow is protected.

Empirical studies related to this theory have examined the hypothesis that highgrowth firms should benefit the most from hedging if the benefit comes from protecting investment funds. The market-to-book ratio has been used as a proxy for the firm's investment demand and is predicted to be negatively related to derivative usage.

In terms of the Froot et al. (1993) intuition, the market-to-book ratio captures the likelihood that future positive NPV projects will be accepted by the firm. If this is the case, then the market value of the high market-to-book firm already includes acceptance of future positive NPV projects. If hedging adds value by increasing the likelihood that positive NPV projects will be accepted, then hedging should be associated with firms that have lower market-to-book ratios. This would logically follow because the market value of low market-to-book firms does not include the acceptance of future positive NPV projects. When examined alone, the evidence on the relation between derivative use and the market-to-book ratio does not support the investment protection hypothesis. Geczy, Minton, and Schrand (1995,1996) do find evidence that firms that are both highly leveraged and have low market-to-book ratios use derivatives more than other firms.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

[&]quot;The concavity assumption means that as the variance in internal funds increases, the profits decrease or: p=f(w); $p_w > 0$, $p_{ww} < 0$, where p is defined as profit and w is defined as internal wealth. Froot et. al point out that this type of assumption is necessary for any model where hedging increases the value of the firm.

Their conclusion is that this finding supports the Froot et al. (1993) model of corporate hedging. This cited prior research looked at market-to-book ratios at one point in time (post-hedging in the case of Geczy, Minton, and Schrand (1995)). Perhaps a better test of the investment protection hypothesis would look for market-to-book ratio changes after derivatives are put in place to see if the value of the firm reflects a higher likelihood of future positive NPV project acceptance.

Another proxy for high growth firms is the firm's relative investment in research and development (R&D). Firms that invest heavily in R&D are predicted to be more likely to use derivatives to protect the internal funds necessary for R&D programs. In empirical studies, R&D usage is consistently significantly related to derivative usage (Nance, Smith, and Smithson (1993), Geczy, Minton, and Schrand (1995)). These results were especially strong for foreign currency derivatives but were weaker for interest rate derivatives, as noted by Fenn, Post, and Sharpe (1996).

2.3 Summary

When considered in total, the empirical evidence to date seems most supportive of the investment protection hypothesis as the primary motivator for derivative usage. This hypothesis provides a value relevant role for hedging cash flows at the firm level.

Therefore, it is assumed for the remainder of this dissertation that owners benefit from an effective hedging strategy. The chosen strategy impacts the earnings process and the earnings return relation, so in addition to being value relevant, the hedge is valuation relevant as well. The following chapter discusses the mechanics of using an interest rate swap as part of an interest rate risk hedging strategy.

CHAPTER III

INTEREST RATE SWAPS AS PART OF A HEDGING STRATEGY

3.1 Introduction

This chapter discusses the practical issues surrounding derivative strategies including the accounting treatment. An example of the mechanics of an interest rate swap is presented, followed by a discussion and example of the accounting treatment. Also discussed are the proposed changes in the generally accepted accounting treatment of interest rate swaps and their impact upon the predictions made in this dissertation.

3.2 Issues surrounding derivative strategies

A derivative instrument is any financial instrument that derives its value from the movement of some underlying asset or index (e.g., stock prices for stock option contracts, foreign exchange rates for currency derivatives, interest rates for interest rate derivatives, etc.). Derivative instruments can increase the exposure to the movement of the underlying index and are sometimes used to "speculate" or exploit expectations about the future direction of movement. Derivative instruments can also decrease the exposure to the movement of the underlying index and are sometimes alternatively used to "hedge" or eliminate the impact of any movement.

Accounting regulators and government officials currently are assessing accounting guidelines for the myriad of derivative strategies adopted by firms. The SEC has recently adopted expanded disclosures about derivative instruments in financial

statements, including qualitative and quantitative disclosures about the market risk of derivative instruments, details of accounting policies for derivative holdings, and the effects of derivatives on other positions. The Financial Accounting Standards Board (FASB) continues its project on derivatives accounting. The U.S. General Accounting Office (GAO) issued a report in May 1994 entitled "Financial Derivatives - Actions Needed to Protect the Financial System". The GAO report found that "accounting standards for derivatives, particularly those used for hedging purposes by end-users, were incomplete and inconsistent and have not kept pace with business practices.

Insufficient accounting rules for derivatives increase the likelihood that financial reports will not fairly represent the substance and risk of these complex activities. In addition, the lack of rules for certain products makes it likely that accounting for these products will be inconsistent, thereby greatly reducing the comparability of financial reports."

(GAO 1994, page 108, para.2) The SEC and GAO proposals highlight that current disclosure practices do not allow users to obtain clear signals about the extent and consequences of derivative use.

The delay in establishing accounting standards for derivative instruments has

⁷The SEC rules are effective for filings after June 15, 1997 for banks, thrifts, and companies with market capitalization over \$2.5 billion. The rules allow companies to choose from among three alternatives for presenting quantitative market risk information that has not previously been disclosed.

^{*}Respondents to the 1995 Survey of Projects and Priorities of the FASB unanimously ranked the project on derivatives and hedging as the top priority for the Board (FASB, Financial Accounting Series, Feb. 26, 1996). The most recent approach discussed by the Board would classify all derivatives on the balance sheet as assets or liabilities and would measure all derivatives at fair value. The accounting for gains or losses on the derivatives would depend upon the reasons for holding the derivative. This approach is presented in an exposure draft dated June 20, 1996. The proposed accounting standards for derivatives would apply to fiscal years ending after December 1997.

occurred because of some fundamental controversial issues regarding the strategies taken by firms in the name of "risk management". The controversy can be broken down into the following three general categories.

First, there was controversy early on in the use of the term "hedging". Some took the view that hedging refers to the elimination of down-side risk only, while others regarded hedging as the elimination of both up-side and down-side risk. The second view, where hedging is the elimination of variance, has generally come to be the accepted definition used by regulators and standard setters.⁹

Second, there is the issue of whether firms are managing entity-wide risk or transaction risk. The managing of entity-wide risk potentially could increase exposure for individual transactions, but the managing of individual transaction risk could increase the exposure of the entity as a whole. For example, an entity-wide strategy of reducing the cash flow impact of interest rate changes may involve changing an existing fixed rate interest expense into a variable rate expense to match existing variable rate income. This action would increase cash flow exposure at the transaction level while reducing cash flow exposure for the entity. Likewise, protecting the interest income associated with a particular investment may increase exposure for the entity as a whole if interest expense is not protected as well.

Third, there is the mutually exclusive nature of cash flow risk management

⁹The latest FAS exposure draft (June 20, 1996) does not explicitly define hedging but the language used ("This statement requires that changes in fair values (or cash flows) of the derivative and hedged item (or hedged transaction) be expected to offset substantially, both at inception and on an ongoing basis." paragraph 135) clearly implies that hedging eliminates both sides of movement.

versus market value risk management. The protection of one necessarily means the exposure of the other. For example, the market value of a variable rate instrument does not change when the interest rate moves but the cash flows associated with the instrument do change. Conversely, the market value of a fixed rate instrument changes when the interest rate changes but the cash flows remain constant. Because of these effects, a derivative strategy of hedging cash flows has opposing impacts on the exposure of market value and a strategy of hedging market values increases the exposure of cash flows.

The task of the regulators and standard setters has understandably been difficult because of these controversies. While most firm managers' report in survey data that hedging of cash flows is their primary reason for using interest rate swaps, the currently required financial statement disclosures may or may not make this clear.¹⁰

This study examines the sample firms' disclosed use of interest rate swaps.

The definition of hedging to be used in the remainder of this study is as follows:

hedging refers to the reduction of variance on an entity-wide level. Because of the

¹⁰In response to <u>The Wharton School/Chase survey of Derivatives Usage Among U.S. Non-Financial Firms</u> (1995), firms responded that 67.2% used hedging strategies to manage cash flow exposure (survey question 5 response). In addition, question 3 of the survey asked firms which derivatives they used to manage interest rate risk and 62.8% of firms responded that interest rate swaps were used to manage interest rate risk.

¹¹The definition used above differs from the most recent FASB exposure draft. The exposure draft applies to *accounting* for hedging at the transaction level. "The board decided that any hedge accounting approach that adjusts the basis of the hedged item could not accommodate a portfolio of dissimilar items (a "macro" approach) because of the difficulties associated with allocating deferrals, changes in fair value,....." (paragraph 161). While the Board expressed a preference for effectiveness assessment (paragraph 163-164) at the entity-wide level, the concerns over the complexity needed to determine the effectiveness of entity-wide hedging activity outweighed the benefits at this time.

cash flows with interest rate swaps and the remainder of the paper evaluates the derivative strategies as cash flow management tools. Because the cash flow effects of interest rate swaps are reflected in earnings, this dissertation uses predictions about the earnings related concepts of persistence and earnings response coefficients to assess the strategy's effectiveness. The next section discusses the mechanics of an interest rate swap and describes how a swap can be used for hedging cash flows. Also described are scenarios where swaps increase the exposure of cash flows to interest rate risk.

3.3 Mechanics of Interest Rate Swaps and Potential Cash Flow Impacts 3.3.1 Example of Interest rate swap mechanics

Interest rate swaps are contracts used to change a variable (fixed) rate cash flow stream into a fixed (variable) rate cash flow stream. An example of the mechanics of a swap helps clarify the transaction.

Suppose Lizco, a solvent company, has variable rate debt (for example LIBOR + .50) but has assets which generate a fixed return (say 10%).¹² Further, suppose Lizco can get a fixed rate of 9.45% in a swap contract. If Lizco enters into the swap contract to swap "variable for fixed", the firm would be obligated to pay interest to the swap counterparty at the 9.45% fixed rate but would be entitled to receive interest from the swap counterparty at the variable rate based upon LIBOR.¹³ As shown in exhibit

¹²LIBOR refers to London Interbank Offer Rate. By convention this is the variable rate used in swap contracts.

¹³The interest to be paid would be calculated based upon an agreed upon "notional" principle, an amount equal to Lizco's variable rate debt if it was a "perfect" hedge. Notional principle is not exchanged

3.1, the net effect of this agreement is that Lizco would incur total net interest expense each period equal to the swap fixed rate of 9.45% plus the .50% spread between the debt variable rate and LIBOR.¹⁴

Exhibit 3.1 Lizco example of interest rate swap mechanics

Lizco example	Cash Flows in(out)
Variable rate on underlying debt	(LIBOR + .50)
Fixed rate in swap agreement	(9.45)
Variable rate in swap agreement	+LIBOR
Net fixed after swap rate	(9.95)
Asset return rate	10.00
Fixed cash spread between assets and liabilities	<u>0.05</u>

From exhibit 3.1, it is evident that Lizco has locked in its cash flow spread by locking the spread between its return on assets and its interest expense. This is one example of how firms can effectively hedge cash flows using interest rate swaps. The interest rate swap contract above insulates the firm's cash flows from changes in the interest rate when combined with the underlying debt contract. It is this combination of the derivative security and the underlying liability that makes the hedge. An interest rate swap could be used to alter the cash flows of either an asset or a liability, but in practice

in cash. The amount is used strictly to compute the amount of interest owed/due under the swap contract.

¹⁴In this example, the original variable rate was based upon LIBOR plus X% and net after swap spread would be constant. If the original variable rate was based upon prime, there could still be some rate variance affected by the correlation of prime and LIBOR rates.

the swap is most often used on the liability side.¹⁵ Therefore, the remainder of this section addresses swaps only on the liability side.

3.3.2 Definition of effective and ineffective cash flow hedging

The effectiveness of a cash flow hedging strategy depends upon the firm's net exposure position before the swap and upon the swap's pay and receive structure. In the prior example, Lizco was in a "net asset fixed" position, where "net asset fixed" means that Lizco's assets are greater than liabilities and that the assets generate a net fixed return. In this position, an effective cash flow hedge must result in fixed rate liabilities. Given the details regarding Lizco's debt, this can only be done with a swap that receives variable rate cash flows and pays fixed cash flows (hereafter referred to as a "swap to fixed" contract). There are four possible pre-swap positions and swap combinations, shown below in exhibit 3.2 along with the hedge effectiveness of each position.

Exhibit 3.2
Effective and Ineffective cash flow hedging

Swap structure: Pre-swap position:	liabilities swap to fixed	liabilities swap to variable
net asset fixed	effective hedge CASE 1	ineffective hedge CASE 2
net asset variable	ineffective hedge CASE 3	effective hedge, CASE 4

¹⁵In the sample of only non-financial firms identified in this study, only one reported that the interest rate swap was used to change the interest rate on an asset and this firm was not used in the analysis due to an unrelated data availability issue.

The effective hedge cases are those where the impact of changes in the interest rate has been eliminated (or reduced) and the ineffective hedge cases are those where the impact of changes in the interest rate has been increased or is unchanged.

If the information necessary to categorize firms into one of these four cases was readily available, financial statement users could easily adjust their expectations of firm future cash flows. Current reporting requirements do not address the net asset position, however, so the categorization is not apparent from reading the financial statements. In order to empirically differentiate among these cells, a proxy for net asset position will be used to partition firms into two groups - one where the swap appears consistent with hedging interest rate risk and the other group where the swap does not appear to be consistent with hedging interest rate risk. The changes in earnings persistence and ERCs coincident with the use of interest rate swaps will be examined for each group.

As noted earlier, the earnings pattern of the firm will reflect the cash flow impact of an interest rate management strategy. The next section describes the current accounting treatment for interest rate swaps. It provides an understanding of how the swaps are recognized in the financial statements and then presents the disclosure requirements for swap activity.

3.4 Accounting treatment for interest rate swaps 3.4.1 Current accounting treatment

The accepted accounting treatment for interest rate swaps was developed by analogy to existing guidance for loosely related financial instruments. In 1984 and 1985, the Emerging Issues Task Force (EITF) discussed several issues relating to

interest rate swaps, thus forming the extent of the "authoritative literature" for these instruments. The issues discussed were hedge criteria and swap termination treatment. These discussions relied upon analogy to FAS Statement 80 and upon developing practice as the basis for comments on the accounting treatment.

FAS Statement 80 "Accounting for Futures Contracts" outlines the proper treatment for exchange-traded futures contracts. This statement sets up the criteria used to allow for hedge accounting, primarily that the contract be designated and effective as a hedge of certain entity-wide exposures. The effectiveness of a hedge is evidenced by the correlation between changes in market value of the hedging contract and the hedged item. If the hedge criteria are met, then changes in the market value of the hedging contract need not be recognized in the period of change. Rather, the change is an adjustment to the carrying amount of the hedged item (as an offset for the change in market value of the hedged item). Essentially this means no income statement effect for changes in market value of the hedging contract.

FAS80 also describes the accounting at termination for futures hedges where any gain or loss on the hedging contract at termination is deferred and recognized when the offsetting gain or loss is recognized on the hedged transaction. An odd result of this treatment is the recognition of an asset (deferred loss) or a liability (deferred gain) at the termination date relating to the hedging item that had not previously been recognized.

Using FAS80 as an authoritative backdrop, the Emerging Issues Task Force provided the following guidance for interest rate swaps in particular.

EITF 84-7 suggests that swap terminations follow the guidance of FAS Statement 80 if the swap had been accounted for as a hedge. The Task Force recognized that there was no consensus for treatment of swaps that had not been accounted for as a hedge.

EITF 84-36 suggests that where there is an underlying debt obligation on the balance sheet, the company should account for the swap as a hedge of the underlying debt and adjust interest expense accordingly. The Task Force recognized that there was diversity in practice in cases where there is no underlying debt or asset associated with the swap transaction.

This is the extent of authoritative guidance on the financial statement recognition of interest rate swaps. The disclosure requirements for interest rate swaps have evolved as part of the Financial Accounting Standard Board's continuing project on derivatives and hedging. Prior to fiscal years ending after June 15, 1990, there was no required disclosure for interest rate swap activity. The earliest statement, FAS 105 "Disclosure of Information about Financial Instruments with Off-Balance-Sheet Risk and Financial Instruments with Concentrations of Credit Risk" required a minimal amount of disclosure regarding a Company's involvement in derivative instruments. The requirements included disclosure of:

- 1) the notional amount of the swap agreement,
- 2) the nature and terms of the swap agreement, with discussion of credit risk, market risk, cash requirements, and related accounting policies,
 - 3) the amount of any accounting loss that would occur as a result of

counterparty default,

4) the policy for requiring collateral and description of any collateral held in connection with the swap agreement.

The disclosure requirements were amended by FAS 107, "Disclosures about Fair Value of Financial Instruments". This statement requires that firms disclose the fair value of financial instruments, including those not recognized in the statement of financial position. This statement allows for a combined presentation of fair values (netting unrealized gains and losses across classes of derivatives).

FAS105 and parts of FAS107 were superseded by FAS 119 "Disclosure about Derivative Financial Instruments and Fair Value of Financial Instruments". FAS 119 specifies that the entity distinguish between derivatives held for trading purposes and those held for purposes other than trading. For derivatives held for trading purposes, the entity must disclose the average fair value and the net trading gains and losses over the reporting period. For derivatives held for other purposes (including interest rate swaps used for hedging) firms must disclose the purpose for holding the instrument and how the instrument is reported in the financial statements. FAS 119 amends FAS 107 to require that classes of derivatives be reported separately and that there be no netting of positions. FAS 119 encourages but does not require quantitative information about market risks.

3.4.2 Example of financial statement recognition and disclosure

Continuing with the Lizco example from exhibit 2.1, the accounting treatment for this particular swap is described below. Suppose that the underlying variable rate

debt has a face amount of \$10,000 and that the notional principal specified in the four year interest rate swap agreement is \$10,000 as well. For purposes of the example, assume that interest is paid annually on both the underlying debt and under the swap agreement and that the variable rate is set as of the beginning of each year. Interest rates over the life of the swap are as follows:

Exhibit 3.3
Lizco Interest Rates

1/1/x1 1/1/x2 1/1/x3	<u>LIBOR</u> 9.45% 10% 11%	Debt variable rate(LIBOR + .50) 9.95% 10.50% 11.50%
1/1/x4	10%	10.50%

If the swap is entered into on 1/1/x1, and is a fairly priced swap, then on 1/1/x1 there would be no cash exchanged and there would be no journal entry made because of the expectation of complete offset of future cash flows (i.e. the receivable completely offsets the payable). Exhibit 3.4 presents the accounting entries that would be made at year-end to recognize the interest expense associated with the underlying debt and to record the interest rate swap activity.

¹⁶FASB Interpretation No. 39, "Offsetting of Amounts Related to Certain Contracts", describes the conditions that must be met to allow for offsetting of amounts in the statement of financial position. Specifically addressed is the applicability to interest rate swaps.

Exhibit 3.4
Accounting entries related to Interest Rate Swaps

At 12/31/x1, interest payable on the debt would be recorded as follows:

Interest Expense

995

Interest Payable

995

To record interest of LIBOR + .50% for the year.

The entries necessary because of the swap agreement would be:

Interest Expense

945

Interest Payable

945

To record the "pay fixed 9.45%" side of the swap.

Interest Receivable

945

Interest Income

945

To record the "receive LIBOR" side of the swap.

As a result of these entries, the financial statements would be impacted in the first year as shown in Exhibit 3.5. As intended by Lizco, the net interest expense shown in the financial statements is at the effectively fixed rate of 9.95% of the debt face amount of \$10,000. For each of the following three years, the accounting entries would be made in a manner as shown in Exhibit 3.4, based upon the rates shown in Exhibit 3.3.

Exhibit 3.5
Financial Statement Impact
12/31/x1

Income Statement dr(cr):	
Interest expense	1,840
Interest income	<u>(945)</u>
Net interest expense	<u>995</u>
Balance Sheet dr(cr)	
Interest receivable	945
Interest payable	(1,840)
Net balance sheet impact	<u>(995)</u>

Exhibit 3.6
Financial Statement Impact
Next three years

	12/31/x2	12/31/x3	12/31/x4
Income Statement dr(cr): Interest expense Interest income Net impact	1,995 (1,000) <u>995</u>	2,095 (1,100) <u>995</u>	1,995 (1,000) <u>995</u>
Financial Statement dr(cr): Interest receivable Interest payable Net impact	1,000 (1,995) (995)	1,100 (2,095) (995)	1,000 (1,995) (<u>995)</u>
On 1/1/X following: Net cash dr(cr):	<u>(995)</u>	<u>(995)</u>	<u>(995)</u>

For the next three years, the financial statement impact would be as shown in Exhibit 3.6. It is clear from this example that the swap has successfully changed the

interest expense to a fixed amount. As shown in Exhibit 3.6, the cash is also a fixed net outflow over the life of the swap.

The disclosure made by Lizco under FAS 105 would be similar to that shown in Exhibit 3.7.¹⁷ After adoption of FAS 107 and FAS 119, the disclosure would include the fair value at year-end of derivative instruments. In accordance with these statements, the fair value would be based upon a quoted price of a similar instrument or a current replacement cost.

Exhibit 3.7 Lizco Disclosure

During 19X1 Lizco entered into a four-year interest rate swap agreement with a commercial bank which effectively fixes the interest rate on its outstanding variable rate debt. Under the agreement the Company pays interest on a \$10,000 notional principal amount based on a fixed rate of 9.45%. The company receives interest based on LIBOR. The interest rate swap was designated as a hedge contract against interest rate fluctuations and accordingly, the differential to be paid or received is accrued as interest rates change and is recognized over the life of the agreement as an adjustment to interest expense. The company has limited exposure to credit loss for the differential between interest rates in the event of nonperformance by the other parties.

 $^{^{17}}$ The wording of this example is based upon the wording found in the disclosures used in this study.

3.4.3 Proposed changes to accounting treatment of interest rate swaps

The example above describes the accounting treatment in place during the time period of the current study. The accounting treatment changes dramatically for interest rate swaps under the exposure draft currently being deliberated . The most important change is a requirement that all derivative instruments be recognized on the balance sheet at fair value. If the derivative is a hedge of the fair value of a particular asset or liability, then that asset or liability is also adjusted to fair value at the balance sheet date. The FASB has also decided that the accounting hedge criteria applies at the transaction level and that the treatment of unrealized gains and losses differs according to the type of hedge designated by management.

For the first time, this exposure draft treats a hedge of market value of an existing asset/liability differently than a hedge of cash flows for a projected transaction. This is important in the area of interest rate swaps because the board has applied this difference such that a swap-to-pay-fixed contract is treated as a hedge of cash flows while a swap-to-pay-variable contract is treated as a hedge of market value.

For hedges of cash flows, the unrealized gains and losses flow through comprehensive earnings until the actual date of the underlying projected transaction. If the hedge is effective then recognition of the swap gain or loss at the date of the transaction should be offset by the gain or loss on the actual transaction.

For hedges of market value, the unrealized gains and losses on the swap contract

¹⁸The exposure draft is dated June 20, 1996. A final statement is expected to be released in June of 1997.

flow through earnings in the current period. If the hedge is effective, these unrealized gains and losses should be offset by the changes in the market value of the underlying asset or liability. Exhibit 3.8 summarizes the treatment for Lizco under the proposed standard. These new rules make the accounting more transparent but do not change the financial statement net effect for successful hedges.

Exhibit 3.8
Financial Statement Impact
Next three years
Proposed accounting treatment

Income Statement dr(cr):	12/31/x2	12/31/x3	12/31/x4
Variable rate debt: Interest expense	1,050	1,150	1,050
Interest rate swap:			
Deferred gains from other comprehensive			
income	(55)	(155)	_(55)
Net impact	995	995	995
Balance Sheet dr(cr):			
Variable rate debt	\$10,000	\$10,000	\$10,000
Interest rate swap	137	265	50
Net payable	(955)	(955)	(955)
Other comprehensive			-
income	(137)	(265)	(50)
On 1/1/Xfollowing:			İ
Net cash dr(cr):	<u>(995)</u>	<u>(995)</u>	<u>(995)</u>

3.5 Summary

This dissertation examines one type of derivative instrument, the interest rate swap, as a tool for management of cash flow exposure to interest rate changes. The economic impact of the interest rate swap agreement on the firm's cash flows has been demonstrated through an example of the interest rate swap mechanics. This chapter has also demonstrated that the accounting for interest rate swaps used by firms during the period of the current study allowed for a matching of the economic consequences of a hedging strategy with the reported accounting earnings since the cash flow effects (i.e., the fixing of cash flows in the example) were identical to the earnings effects (i.e., the fixing of interest expense).

This chapter provides the background for understanding the valuation relevance of a particular type of hedging strategy. The earnings pattern of the firm is altered by the hedging strategy and accordingly, the financial statement users' reaction to earnings news is affected. The following chapter develops predictions regarding the valuation relevance of an interest rate swap strategy.

CHAPTER IV

THE IMPACT OF A DERIVATIVE STRATEGY ON THE EARNINGS PROCESS

4.1 Introduction

This chapter develops predictions regarding the impact of an interest rate swap strategy on the earnings process. Predictions are made regarding the persistence of surprises in the earnings process and regarding the earnings/return relation.

4.2 The valuation relevance of the earnings stream

Cash flow hedging impacts the earnings stream of the firm and the relation between stock returns and earnings has been the focus of many studies. One conclusion of this literature is that investors use earnings information to update their expectations of future cash flows. A theoretical model used in many of these studies is the traditional formulation of price as the present value of expected future cash flows. Assuming that the present value of revisions in expected future cash flows is equivalent to the present value of revisions in the expected future earnings, Kormendi and Lipe (1987) show that the unexpected change in price is a function of unexpected earnings. When unexpected returns are regressed on unexpected earnings, the coefficient on the unexpected earnings is the "earnings response coefficient" or ERC.

Kormendi and Lipe (1987) provide empirical evidence that ERCs vary cross-

¹⁹The work of Miller and Rock (1985) first described the earnings announcement effect. Other works in this area include Kormendi and Lipe (1987), Easton and Zmijewski (1989), Collins and Kothari (1989), Salamon and Stober (1994), and others.

sectionally depending upon the earnings process of the firm. The earnings process is a function of the nature of the business, the operating environment, investments made, and the form of contracts with third parties (such as suppliers, managers, debt investors. employees, etc.). Actual earnings is generally modeled as a process that has an expected change and a variance around the change (due to risk in investments and business contracts and due to unexpected changes in the operating environment). Earnings surprise is the difference between the actual earnings change and the expected earnings change. Earnings persistence captures the extent to which earnings surprises persist into future periods. Prior research implies that if persistence is higher, then investors should update their estimate of expected future cash flows more for a dollar of earnings surprise than if persistence is lower.²⁰ Within the framework of Kormendi and Lipe (1987) and other prior research, the persistence of innovations in earnings depends upon the form of the model of the actual earnings process. Most prior studies have modeled the persistence of the annual earnings process, but because short-term interest rates are used in interest rate swap contracts, a model of the persistence in the quarterly earnings process is more useful for generating empirical predictions in this study.

4.3 A model of persistence in the quarterly earnings process

Earnings is exposed to various risks (for example, interest rate risk, exchange rate

²⁰Miller and Rock (1985) model the effect of earnings announcements upon prices and the model includes an earnings persistence parameter. Kormendi and Lipe (1987) use the Miller and Rock model to develop an earnings persistence measure that is used to explain cross-sectional differences in earnings response coefficients (ERCs). Kormendi and Lipe provide empirical evidence that cross-sectional differences in ERCs are related to cross-sectional differences in earnings persistence. Easton and Zmijewski (1989) and Collins and Kothari(1989) also find evidence that ERCs are positively related to earnings persistence.

risk, product supply and demand risk, etc.) and earnings is correlated with measures that capture these risks. This study focuses on one type of risk - interest rate risk. If some component of earnings is correlated with short-term interest rates, then that earnings component will be described by the time series of interest rates.

Suppose C_t is the component of net earnings that is dependent upon interest rates. C_t includes not only those items explicitly dependent upon interest rates such as variable rate interest income and expense (A_t) but also includes the portion of all other income and expense items implicitly related to the short-term structure of interest rates (B_t) .

$$C_t = A_t + B_t \,, \tag{1}$$

$$A_t = DR_t, \text{ where} ag{1.1}$$

D = the carrying value of the net assets that are explicitly interest sensitive, and

 $R_t =$ the short-term interest rate;

$$B_t = \gamma_t + KR_t, \text{ where}$$
 (1.2)

 γ_t and K describe the linear relation between the implicitly interest sensitive earnings components and the short-term interest rate. Then the realization of C_t can be described with the following model:

$$C_{t} = DR_{t} + \gamma_{t} + KR_{t}, \text{ or}$$

$$C_{t} = \gamma_{t} + (D+K)R_{t}$$
(1.3)

The short-term interest rate is examined by Brenner, Harjes, and Kroner (1996) (BHK)

who present the following model^{21,22}:

$$R_{t} = R_{t-1} + \alpha - \beta R_{t-1} + \epsilon_{t}$$

$$= \alpha + (1 - \beta)R_{t-1} + \epsilon_{t},$$
(2)

where R, is the short-term interest rate at time t,

 α is a constant drift term.

(1- β) is an autoregressive parameter, and $\beta \ge 0$, and

 ϵ_{t} is the realization from a random process with $E(\epsilon_{t})=0$ and $Var(\epsilon_{t})=\sigma^{2}_{\epsilon}$. This model describes a process that is autoregressive in levels, with $(1-\beta)$ as the autoregressive parameter. The size of the coefficient β determines the speed of the decay of R_{t-1} in future levels of R^{23} . For example, if $\beta=0$ in the above process, then R_{t-1} remains fully in the future levels of R and the process would be a random walk with drift. If $\beta=1$ then R_{t-1} has no impact on future levels of R and the process could be described as completely reverting to the drift term, α . Substituting the model of R_{t} from equation (2) into equation (1.3),

$$C_t = \gamma_t + (D+K)(\alpha + (1-\beta)R_{t-1} + \epsilon_t)$$
(3)

²¹This model is the discrete analogy to the continuous time models of Cox, Ingersoll, and Ross 1980) and others.

 $^{^{22}}$ BHK are primarily concerned with modeling the volatility of rates through specification of the conditional variance of ϵ_t , which is not examined here. They point out that the model above is useful because of its generality and that other models exist that may better describe the quarterly path of interest rates. The model above is used here because of its familiarity and because it is adequate for creating insights about the possible behavior of earnings before and after the initiation of interest rate swaps.

 $^{^{23}}$ BHK examined weekly rate changes and did not find significance for the β parameter over these short intervals, but the reversion of rates around a long-term mean level is an accepted premise in the term structure literature. For example, Chan, Karolyi, Longstaff, and Sanders (1992) empirically estimate parameters over 1 month intervals for the above discrete model under 9 alternative specifications and find that the parameter estimates for β range from .1779 to .5921 for models that explain up to 20% (when β =.5921) of the volatility of changes in interest rates (table 3, pg.1218).

where all items are as described above.

The C_t process follows the same AR process as the interest rates. Again the size of the β coefficient in the AR parameter (1- β) determines the speed of decay of prior realizations in future levels of the series. It is this property of decay in future levels that is key in evaluating the persistence of surprises in this component of earnings relative to the other earnings component discussed below.

The portion of earnings that is <u>not</u> correlated with interest rates is assumed to follow the quarterly seasonal Brown and Rozeff (1979) model²⁴. The following standard form of the model applies:

$$S_{t} = S_{t-4} + \phi(S_{t-1} - S_{t-5}) + \theta_{4}a_{t-4} + a_{t}, \qquad (4)$$

where S_t is the quarterly level for the present period, t, of the earnings component not correlated with the interest rate process,

 ϕ is an autoregressive parameter and $0 < \phi < 1$,

 θ_4 is a moving average parameter and $-1 < \theta_4 < 0$,

 a_t is the realization from a random process with $E(a_t) = 0$ and $Var(a_t) = \sigma^2$, and $corr(S_t, C_i) = 0$ for all t, j.

The quarterly earnings for the firm, QE, is a combination of the C and S series:

$$QE_t = C_t + S_t. (5)$$

Substituting from equations(3) and (4) into equation (5) yields

 $^{^{24}}$ Brown and Rozeff(1979) find that the movement in quarterly earnings is reasonably well explained by a seasonal time series model of the form (1,0,0)x(0,1,1). This model was found to explain roughly 75% of the variation in quarterly earnings in their sample of firms. This model was also used by Bernard and Thomas(1990) as a model for the quarterly earnings process.

$$QE_{t} = \gamma_{t} + (D+K)(\alpha + (1-\beta)R_{t-1} + \epsilon_{t}) + S_{t-4} + \phi(S_{t-1} - S_{t-5}) + \theta_{4}a_{t-4} + a_{t}, (5.1)$$

where all terms are as described above²⁵.

As described by Enders (1995), equation 5.1 can be decomposed into a deterministic trend in the γ_t and $(D+K)\alpha$ terms and a stochastic trend in the $\{(D+K)((1-\beta)R_{t-1}+\epsilon_t)+S_{t-1}+\epsilon_t\}$ term. If thought of as the permanent portion of the $(D+K)\epsilon_t$ and a_t realizations, the stochastic trend can show how the earnings surprises $((D+K)\epsilon_t$ and a_t) are carried into the future periods beyond t.²⁶ Splitting the QE_t series into its independent component parts, C_t and S_t , the permanence factor, Ψ_{ct} , for the C_t series (the expectation at time t of how the realization of $(D+K)\epsilon_t$ is carried forward j periods) can be expressed as:

$$\Psi_{ct} = (1 - \beta)^{j} (D + K) \epsilon_{t}, \text{ for all } j$$
 (6)

where the terms are as described above.²⁷ As noted earlier, the size of the β parameter determines how the current period surprise decays in future periods. It is apparent from equation (6) that the impact of $(D+K)\epsilon_t$ approaches zero over subsequent future periods.

In a similar way, the permanence factor, Ψ_{st} , for the S_t series can be shown (the expectation at time t of how the realization of a_t is carried forward j periods into the future) as:

$$\Psi_{st} = \phi^{j} a_{t}, \qquad j < 4, \qquad (7)$$

²⁵The simple combination of the two series relies upon two attributes:: the additive property of normally distributed random variables and the assumed independence of the two series.

²⁶The expression of the permanence factor is similar to those found in Table 1 of Collins and Kothari(1989), who summarize permanence factors of various annual earnings time series models.

 $^{^{27}}$ Appendix A shows the derivation of the permanence factor Ψ_c .

$$\Psi_{st} = \Phi^{j-4} a_t + \Phi^{j} a_t + \Phi^{j-4} \theta_4 a_t \qquad 4 \le j < 8,$$

$$\Psi_{st} = \Phi^{j-8} a_t + 2\Phi^{j-4} a_t + \Phi^{j} a_t + \Phi^{j-8} \theta_4 a_t + \Phi^{j-4} \theta_4 a_t \qquad 8 \le j < 12, \text{ etc..}^{28}$$

The moving average and seasonal autoregressive parameters in the S₁ series cause the current period surprise, a₁, to be carried to future periods.

The persistence for each of these series, C_t and S_t , is the present value of the sum of the Ψ_{ct} and Ψ_{st} over all future j periods.²⁹ The persistence for the Ψ_{ct} series is described as:

$$PV\Psi_{ct} = (D+K) \in \sum_{t=1}^{\infty} \frac{(1-\beta)^t}{(1+b)^t} = (D+K) \in \frac{(1-\beta)}{b+\beta},$$
 (8)

where b is the assumed discount rate and all other terms are as described. The persistence for the Ψ_{st} series is more complicated because of the seasonality, but basically can be described as:

$$PV\Psi_{st} = a_{t} \sum_{j=1}^{\infty} \frac{(1 + \phi' + \phi'\theta_{4})}{(1 + b)^{j}}$$
$$= a_{t} (\frac{1}{b} + \frac{\phi + \phi\theta_{4}}{1 + b - \phi}), \tag{9}$$

where all terms are as described earlier³⁰. The persistence of QE_t is dependent upon the proportions of C_t and S_t and the relative persistence of Y_{ct} versus Y_{st} . Comparing the persistence in terms of the time series parameters only:

²⁸Appendix A shows the derivation for Ψ_s .

²⁹This definition of persistence is analogous to that used by Kormendi and Lipe (1987), "the present value of the revisions in expected future earnings induced by a \$1.00 innovation in current earnings, is our measure of persistence".

³⁰Equation 9 is a simplification of the S series where all of the 4th quarter effects are combined.

$$\frac{1-\beta}{b+\beta} < (\frac{1}{b} + \frac{\phi + \phi \theta_4}{1+b-\phi}), \tag{10}$$

or

$$PV\Psi_{ct} < PV\Psi_{st}$$

where $0<\beta<1$, $0<\phi<1$, $-1<\theta<0$, and 0<b<1.

The total persistence for QE_t is a weighted average of the persistence of the two series, with the weights coming from the absolute magnitude of the errors. This total can be described as:

$$PV\Psi_{ct}\left(\frac{(|D+K|)|\epsilon_t|}{|(D+K)||\epsilon_t|+|a_t|}\right)+PV\Psi_{st}\left(\frac{|a_t|}{(|D+K|)|\epsilon_t|+|a_t|}\right), \tag{11}$$

Examination of equations (10) and (11) indicates that the persistence of the QE_t earnings stream is a weighted average of the persistence of the C_t and S_t series and is dependent upon the β , ϕ , θ , and (D+K) parameters. As noted earlier, the β parameter comes from the interest rate series and is exogenous to the firm. The ϕ and θ parameters in the QE_t series described above are combinations of exogenous risks and management choices not specified in this study. The (D+K) parameter is the amount of explicitly interest sensitive net assets (K) and the amount of implicitly interest sensitive net assets (K). Referring back to the example of the mechanics of an interest rate swap in Chapter 3, it is clear that the variable rate side of the swap follows the same short term interest rate process, R, described above. The mechanics of the swap by design lead to a change in the amount (D+K) of interest sensitive net assets (because of the use of the notional amount). It is then the (D+K) parameter that is impacted when the firm implements an interest rate derivative strategy and it is through the (D+K) parameter that the derivative

strategy affects the persistence of the earnings stream. Exhibit 4.1 shows the persistence of the total earnings under a range of values for the (D+K) parameter.

Exhibit 4.1 Values of Persistence under different |D+K| parameter values

Total Persistence	PVΨ,	PVΨ _c	b discount rate 0≤K≤1	β int. rate speed 0≤β≤1	φ seasonal ar 0 <φ<1	θ seasonal ma ³¹ -1<θ<0	D+K value	€, value ³²	a value
9.29	82.0	0.75	0.10	0.50	0.50	-0.70	90.00	0.01	8.00
9.49	82.0	0.58	0.10	0.50	0.50	-0.70	70.00	0.01	8.00
9.69	82.0	0.42	0.10	0.50	0.50	-0.70	50.00	0.01	8.00
10.13	82.0	0.08	0.10	0.50	0.50	-0.70	10.00	0.01	8.00
10.25	82.0	0.00	0.10	0.50	0.50	-0.70	0.00	0.01	8.00

4.4 Interest rate derivative strategies and persistence

From an inspection of equation (11) and exhibit 4.1, it is evident that as |D+K|-0, the weight of the C_t series decreases. Eliminating the less persistent earnings component leads to an increase in the persistence (because the more persistent series has a greater weight in the weighted average). If interest rate derivatives are viewed as altering (D+K) then when D+K=0 the firm has completely hedged its cash flows against the interest rate risk. If the firm took a position inconsistent with cash flow hedging, then the interest sensitivity would not be decreased and interest rate changes would remain (or

 $^{^{31}\}text{The parameter values of .5 for }\varphi$ and -.7 for θ come from an example in Brown and Rozeff (1979).

³²The value of $(D+K)\epsilon$ is chosen arbitrarily but within reason relative to the a value chosen, given that the interest rate sensitive component of most non-financial firms is material but is dominated by the non-interest rate sensitive component.

be magnified) in earnings changes and the |D+K| value would not move towards zero. Thus, the impact of alternative derivative strategies upon persistence in the aggregate earnings stream differs for each type of strategy.

Referring back to Exhibit 3.2 from Chapter 3, the four cases can be examined in light of this persistence discussion. In both Cases 1 and 4, where there is an effective cash flow hedge, the strategy reduces the amount (D+K-0) of interest sensitive net assets. This leads to an increase in the permanence of the earnings surprises for Case 1 firms and Case 4 firms.

On the other hand, consider cases 2 and 3 where the strategy is inconsistent with effective cash flow hedging. The ineffective interest rate strategy maintains or increases the amount (D+K does not go to 0) of interest sensitive net assets. This leads to no change or a decrease in the permanence of the earnings surprises for Case 2 firms and Case 3 firms.

The conflicting implications for persistence from swaps that effectively hedge versus those that do not highlight the mutually exclusive affects of the different strategies on earnings persistence. The discussion above leads to the prediction that persistence will increase after an effective cash flow hedge of interest rate risk and will not increase after a position is taken that is inconsistent with an effective cash flow hedge. The first part of this dissertation's empirical tests investigates whether interest rate swaps are material enough to change the earnings persistence of a group of firms that are believed to enter into swaps as part of a cash flow hedging strategy. Furthermore, the direction of change in persistence is used as an indicator of whether the consequences of the swaps for this

group are more consistent with the stated cash flow hedging strategy or with a strategy that is inconsistent with cash flow hedging.

4.5 Interest rate derivative strategies and earnings response coefficients

The second part of this dissertation investigates whether ERCs are impacted by the change in earnings persistence that comes about from initiation of an interest rate derivative strategy. Prior research has documented that ERCs are directly related to the earnings persistence. However, ERCs are affected not only by the persistence, but also by the shareholder's perceived riskiness of the firm (Easton and Zmijewski (1989)), growth rates and interest rate levels (Collins and Kothari (1989)), and the perceived quality of the reported earnings (Collins and Salatka (1993)) among other factors.

The riskiness of the firm has been found to be inversely related to ERCs (see Easton and Zmijewski (1989), Collins and Kothari(1989)). While the preceding model focuses upon persistence and no predictions are made about the strategy's impact upon the riskiness of the firm, prior research in the area of derivatives has not found a significant change in market betas for firms following the implementation of a derivative strategy (Hentschel and Kothari 1995). Hentschel and Kothari (1995) interpreted this finding of no change as support for the use of derivatives as a risk management tool rather than as a gamble by managers because the riskiness did not increase significantly for firms in their sample. The intuition of their hypothesis is that a speculative derivative strategy should be associated with an increase in riskiness. Because changes in risk are inversely related to ERCs, a(n) decrease (increase) in the perceived riskiness of the firm following the implementation of a derivative strategy would be expected to be associated with

increased (decreased) ERCs. Therefore, the Hentschel and Kothari intuition regarding riskiness would result in predictions about ERCs identical to the predictions based upon the ERC/persistence relation described above. This dissertation therefore attempts to control for risk as a possible confounding factor in the empirical analysis.

Growth rates have been identified as having a direct relation to ERCs (Collins and Kothari (1989)). This dissertation makes no predictions about the relation between growth rates and persistence. If the implementation of a derivative strategy coincides with a change in the growth rate of the firm, then changes observed in ERCs may be attributable to the growth rate changes. Therefore, growth rates represent another possible confounding factor that is controlled for in the empirical analysis of this study.

Interest rate levels have been found to be inversely related to ERCs (Collins and Kothari (1989)) over time. This dissertation makes predictions based upon the correlation between the earnings stream and the term structure of interest rates, an issue separate from the relation between ERCs and the level of interest rates across time. In the formulation of the earnings stream found in equation (5.1), the changes in interest rate levels over time would be captured by the α and β parameters, both exogenous to the firm. Persistence would be impacted not by a change in the level of the interest rate, but only by a change in the size of the β parameter (the speed of decay parameter). Such an exogenous change would have a market wide impact, not only for firms with interest rate strategies but also for firms with no strategies in place. The theory behind the interest rate level/ERC relation stems from the discount rate used to capitalize earnings changes over time. This rate is theorized to move intertemporally but is not expected to vary

cross-sectionally. In this dissertation, a control firm sample is used to capture macro changes across time, beyond the treatment effect, and the interest rate level movement represents one such macro change.

Collins and Salatka (1993) find an inverse relation between ERCs and the perceived quality of the earnings report. The impact on the ERC from the current accounting practices of reporting derivatives activity is difficult to predict ex ante.

Therefore, the part of this research that attempts to discover whether the initiation of interest rate derivatives is associated with changes in ERCs assumes that there is no significant change in perceived quality of the earnings report upon implementation of an interest rate derivative strategy.

Specifically, the second part of this research investigates whether the earnings response coefficients change for a group of firms believed to have implemented an interest rate hedging strategy. The direction of change in the ERCs is predicted to be the same as that predicted for the persistence and is used as an indicator of whether the consequences of the swaps for this group are more consistent with the stated cash flow hedging strategy or with a strategy that is inconsistent with cash flow hedging. Exhibit 4.2 re-creates exhibit 3.2 and adds the persistence and ERC predictions discussed above.

Exhibit 4.2 Predictions for persistence and ERCs following implementation of strategy

Swap structure: Pre-swap position:	Liabilities swap to fixed	Liabilities swap to variable		
Net asset fixed	effective hedge CASE 1 ΔPersistence >0 ΔERC > 0	ineffective hedge CASE 2 △Persistence ≤ 0 △ERC ≤ 0		
Net asset variable	ineffective hedge CASE 3 ∆Persistence ≤ 0 ∆ERC ≤ 0	effective hedge CASE 4 ΔPersistence >0 ΔERC >0		

4.6 Summary

This chapter developed predictions for persistence and ERCs following implementation of an interest rate swap strategy. These predictions are modeled and tested empirically in Chapter VI (Empirical tests of the persistence predictions) and Chapter VIII (Empirical tests of the ERC predictions). The following chapter describes the sample used for the empirical tests.

CHAPTER V

SAMPLE SELECTION, CATEGORIZATION OF FIRMS, AND DESCRIPTIVE STATISTICS

5.1 Introduction

This chapter describes the process used to select the treatment and control samples. It also includes information about the classification of firms into groups of "effective cash flow hedging" firms and "ineffective cash flow hedging" firms. Finally, descriptive statistics are presented for the firms used in the empirical tests of persistence and ERC changes after implementation of an interest rate swap strategy.

5.2 Sample selection

The National Automated Accounting Research System (NAARS) was accessed for all years 1972 through the present.³³ A search was made on the string "interest rate swap" for each year, starting with 1992 and working backwards until there were no items found. The earliest reports found were in 1982 when four firms reported the use of interest rate swap agreements. The number of reports identified each year increased rapidly in the subsequent years with 590 firms reporting interest rate swap activity in 1992.

Because this study does not address the derivative strategies of financial institutions, all financial institutions were eliminated from the sample. This was done by

³³The search was made using Lexis/Nexis database.

deleting all firms with four digit SIC codes between 6000 and 7000.

Using the 1992 reports as a starting point, the firms were traced back through NAARS to their earliest disclosure year, which was designated as the identification year. Because of the small number of firms disclosing in pre-1989 years and because of the limited data availability after 1992, the analysis includes only firms that first disclosed swap activity in 1989, 1990, and 1991. Because it is unknown how long the swaps were in place during the identification year, the first month of the year immediately following the identification year was used to divide a quarterly time-series of firm quarterly earnings forecast errors, earnings forecast revisions, and abnormal returns into a pre-disclosure and a post-disclosure period.

The 1994 Compustat Industrial Quarterly P-S-T and Research files were searched to find the following for each firm quarter available for five years before the identification year and up to 4.5 years after the identification year: book/market ratio, total market equity, operating income before depreciation, depreciation, SIC code, fiscal quarter, fiscal year end month, and quarterly earnings announcement date.

Using the earnings announcement date from Compustat, the quarterly earnings per share forecast, actual quarterly earnings per share, and quarterly earnings per share forecasts of quarter t+1 were extracted from the Value Line Investment Survey. To be included in the sample, Value Line data must be available for the entire time period (1984-1994). This requirement insured that a continuous time series of observations could be used in the revision test and in the ERC test.

In order to examine cross-sectional differences in the pre- and post-disclosure

periods, control firms were chosen for each treatment firm from Value Line based upon industry and size.

Table 5.1 Sample Selection Details

Sample Selection - Treatment firms:	
NAARS Search of 1992 annual reports ³⁴	590
Eliminate financial services firms	-167
Eliminate firms first swapping in years other than 1989,1990,1991	-155
Eliminate firms not covered by Compustat	-137
Eliminate firms with incomplete data from CRSP or Compustat over the sample period ³⁵	-20
Eliminate firms not covered by Valueline	<u>-19</u>
Total treatment firms	<u>92</u>
Sample selection - Control firms:	
Firms selected by matching on size and industry ^{36,37}	89
Eliminate firms not covered by CRSP or Compustat for the required sample period	<u>-8</u>
Total Control firms	<u>81</u>

A search of annual reports for all control firms insures that no control firm was

³⁴This is the number of firms from the NAARS search for which cusip numbers were found using the following sources: Standard and Poor's CD Corporate data base, CRSP flat file.

³⁵Coverage was necessary over the time period before and after swapping.

³⁶The industry categorization was based upon Valueline's groupings of firms.

³⁷In three cases, two treatment firms were matched to the same control firm. This was necessary due to limits imposed by the matching criteria and by Valueline's coverage of the industry.

disclosing the use of interest rate swaps during the test period. All treatment and control firms must have daily price and return data available on the 1994 version of the Daily CRSP tape for the entire testing period.

5.3 Categorization of firms

5.3.1 Model of interest rate sensitivity

In order to make directional predictions about the persistence and ERCs following the implementation of a hedging strategy, the treatment firms need to be categorized according to the 4 cases shown in Exhibit 4.2. As noted earlier, the financial statements do not explicitly provide the net asset interest rate sensitivity, so a reasonable proxy must be chosen. Because the Exhibit 4.2 cases are differentiated by the sensitivity of net assets to interest rate changes, a proxy based upon the interest rate sensitivity of operating income was deemed appropriate.

For each firm in the treatment group, the following model was used to determine the extent of the relation between changes in operating income and changes in short-term interest rates:

$$Opinc_{qj} = d_0 + d_1 Chgrates_q + d_2 Chgrates_{q-1} + e_q. where$$
 (12)

Opinc_{qj} = the change in operating income after depreciation for firm

j for quarter q, before interest income or interest expense,

Chgrates_q = the change in the 3 month T-bill rate over quarter q,

Chgrates_{q-1} = the change in the 3 month T-bill rate over quarter q-1.

5.3.2 Pre-period interest sensitivity

The model was first run for each treatment firm over the time period before the disclosure of interest rate swap activity. The individual regressions were evaluated to

determine the interest sensitivity of each treatment firm. The results, as shown in Table 5.2 under the "Pre-swapping" column, indicated that for 14 firms the operating income was found to be interest sensitive, as evidenced by a significant coefficient estimate on either the Chgrate_q or Chgrate_{q-1} variable. Significance for this test was determined at the two-tailed 5% level. For these 14 firms, the model was able to explain on average roughly 36% of the change in operating income from quarter to quarter during the preswapping period, indicating a strong relation between changes in interest rates and changes in operating income.

For the remaining 78 firms, the individual regressions indicated no systematic relation between the changes in short-term interest rates and changes in operating income from quarter to quarter. For this group of firms, the model was not able to explain the change in operating income as evidenced by the average adjusted R² of -6% and the low average model F statistic of .5795. The operating income of these firms was therefore designated as interest insensitive over the time period before disclosure of interest rate swap activity.

5.3.3 Post-period interest sensitivity

The model was also run for each treatment firm over the time period after the disclosure of interest rate swap activity. The interest sensitivity of the operating income was evaluated in the same manner and again 14 firms were found to be interest sensitive, as shown in Table 5.2 under the "Post-swapping" column. For these 14 firms, the model was able to explain on average roughly 31% of the change in operating income from quarter to quarter during the post-swapping period.

In the "Post-swapping" period, the individual regressions for the remaining 78 firms indicated no systematic relation between the changes in short-term interest rates and changes in operating income from quarter to quarter. The average R² of -4.3% and the low average model F statistic of .7253 reflect the model's inability to explain the change in operating income for these firms. These firms were accordingly classified as having operating income that was insensitive to changes in short-term interest rates over the period after disclosure of interest rate swap activity.

5.3.4 Decision rule for grouping firms

Table 5.2 summarizes the results for the individual regressions over each of the two time periods. Of the 14 firms found to be interest sensitive in the "Post-swapping" period, only 3 of these 14 are included in the "Pre-swapping" category of interest sensitive. This means that there are 11 firms in the "Pre-swapping" interest sensitive category that become interest insensitive in the "Post-swapping" period. Likewise, there are 11 firms in the "Post-swapping" interest sensitive category that were interest insensitive in the "Pre-swapping" period. For these 22 firms, something happened that was unrelated to the firm's interest rate swap activity.

Because of the inability to isolate the impact of interest rate swap activity from the impact of this unknown operating change, these 22 firms and their matching control firms were eliminated from the empirical tests. After elimination of the 22 firms from the interest sensitive category, there remained 3 treatment firms that were categorized as interest sensitive in both the "Pre-swapping" period and in the "Post-swapping" period.

For the empirical tests, these firms were classified as "net asset variable" firms.

Table 5.2 Summary of Categorization Regression results based upon the interest sensitivity in the pre-swapping quarters and in the post-swapping quarters

Model: Opinc_{qi} = $\alpha_0 + \alpha_1$ Chgrate_q + α_2 Chgrate_{q-1} + ϵ

	Pre-swapping	Post-swapping	
Number of treatment firms in total	92	92	
Average number of observations per regression	14.3	14.2	
Panel A: Firms where operating income was found to b	e interest sensitive		
Number of interest sensitive treatment firms	14	14	
Average number of observations per regression	13.9	14.07	
Average adjusted R ²	36.52%	31.08%	
Average F statistic	5.428	4.195	
Number of times Chgrate _q coefficient positive and significant (negative and significant) ^b	6(4)	1(4)	
Number of times Chgrate _{q-1} coefficient positive and significant (negative and significant) ^{b,c}	6(3)	9(4)	
Panel B: Firms where operating income was found to b	e interest insensitiv	/eª	
Number of interest insensitive treatment firms	78	78	
Average number of observations per regression	14.56	14.35	
Average adjusted R ²	-6.008%	-4.306%	
Average F statistic	.5795	.7253	
Number of times Chgrate _q coefficient positive and significant (negative and significant)	0(0)	0(0)	
Number of times Chgrate _{q-1} coefficient positive and significant (negative and significant)	0(0)	0(0)	

Opinc_q = the change in operating income after depreciation for quarter q, before interest income or interest expense,

Chgrate_q = the change in the 3 month T-bill rate over quarter q,

Chgrate₀₋₁ = the change in the 3 month T-bill rate over quarter q-1.

^{*}Firms were determined to be interest sensitive if the estimated coefficient on Chgrate_{q-1} was significantly positive or negative.

^b Significance was determined at the two-tailed 5% level.

There were 5 pre and 4 post cases where both Chgrate_q and Chgrate_{q-1} were significant, totaling 14 firms.

After elimination of the 22 firms from the interest insensitive category, there were 67 firms that had no evidence of interest sensitivity in either time period. For the empirical tests, these firms were classified as "net asset fixed" firms.

The finding that operating income for most treatment firms is insensitive to changes in the short-term interest rate is not surprising. It is consistent with a sample which does not include financial services industries and is made up primarily of capital intensive industries.

5.3.5 Disclosure

The direction of the interest rate swap was determined for each treatment firm.

The disclosures typically provide an indication of the direction of the swap cash flows.

If the firm discloses that the swap agreement calls for payments based on a fixed rate, then the swap was classified as a "swap-to-fixed" interest rate swap. In this case, cash outflows and interest expense are fixed by the agreement. If the firm disclosed that the swap agreement called for payments based upon a variable index, then the swap was classified as a "swap-to-variable" interest rate swap. Exhibit 1 provides an example of the wording from an actual disclosure.

5.3.6 Final Grouping

Once all treatment firm disclosures had been classified as "swap-to-fixed" or "swap-to-variable", the treatment firms were grouped according to the four cases shown in Exhibit 4.2. This grouping is necessary to test the predictions made for the impact of effective cash flow hedges upon firm persistence and upon the firm's ERC. Table 5.3 shows the final grouping of the treatment firms for the empirical tests. The control

firms were assigned to the same cell as their matching treatment firm.

Exhibit 5.1 Disclosure example "Swapping to fixed"

NOTE-10: [DEBTAC COMMT]

•••••

The company has entered into interest rate swap agreements whereby the company pays interest on the notional amount based on a fixed rate. (italics added) At January 30, 1991, the company had two interest rate swap agreements outstanding with commercial banks, having a total notional principal amount of \$ 150 million. The variable rate is a calculated bond equivalent based on the rate for 30-day commercial paper. These agreements expire during fiscal years 1992-1995. The weighted average effective interest rate on these agreements during 1990 was 7.4%. The company has limited exposure to credit loss for the differential between interest rates in the event of nonperformance by the other parties.

Principal payments on long-term debt for five years subsequent to 1990, in millions, are: 1991-\$ 51; 1992-\$ 37; 1993-\$ 140; 1994-\$ 55; 1995-\$ 233.

NOTE-11: [DEBTAC]

Table 5.3

Categorization of Treatment Firms
to hedging strategy cases

Total number of Treatment Firms = 70

Categorization of Treatment firms ^a						
Swap structure: Pre-swap position:	swap to fixed	swap to variable				
net asset fixed	effective hedge CASE 1 56 firms	ineffective hedge CASE 2 11 firms				
net asset variable	ineffective hedge CASE 3 3 firms	effective hedge CASE 4 0 firms				

^{*}The control firms were grouped into the corresponding treatment firm's cell.

5.4 Statistics for categories

The final sample to be used in the persistence test includes 4,389 quarterly observations, representing 70 treatment firms and 60 control firms. The difference between total treatment and total control is due to 7 control firms that did not have complete information on Compustat or CRSP and 3 control firms acting as the match for more than one treatment firm.³⁸ The total number of quarterly observations excludes quarters where the forecasted earnings was zero, due to the formulation of the variables as percentages of forecasted earnings. Also eliminated from the final sample are quarters where a restructuring charge was taken, since revisions of future earnings was

³⁸Because of the necessity to be followed by Value Line, some control firms represented the only suitable match for more than one treatment firm. Also, because of the need for daily price information, 7 of the control matches chosen were not used. As shown in tables 5.4 and 5.5, the final control and treatment samples were well matched even after the elimination of these 7 firms.

likely for these quarters due to the restructuring, regardless of the derivative strategy.

The total firm quarters used in the ERC test is 4,476, with the increase due to inclusion of the zero forecasted earnings quarters.

5.4.1 Univariate statistics across all time periods

Table 5.4 provides information on the sample on the following dimensions: total assets, market equity, book equity/market equity ratio, book debt/book equity ratio, and β . The statistics were run over all quarters available for each firm. There are no unusual items noted from examination of this table. It appears, by comparison of total assets and market equity, that Case 2 and 3 firms (the ineffective cash flow hedging firms) are significantly larger than the Case 1 and 4 group (the effective cash flow hedging firms), although size appears to be the only significant difference between the two. The impact of this size difference between groups is discussed in the conclusion of this dissertation in Chapter VIII.

5.4.2 Test of means between treatment and control

The matching was effective as evidenced in the comparative statistics for the treatment and control firms. Table 5.5 shows the result of tests of differences in means between the treatment and control firms for size (measured as total equity), book equity/market equity ratio, and β . For this test, the individual firm means were calculated first and then the firm means were used to compute treatment and control firm means for comparison. The t-statistics were calculated assuming unequal variances in all cases where an F-test for equality of variances indicated a rejection of the null hypothesis of equality. The means were equivalent with regard to size and there was no

significant difference between the book equity/market equity ratio or β .

5.4.3 Tests of differences between pre- and post-swapping time periods

Table 5.6 provides an analysis of differences between the pre- and postswapping time periods for treatment and control firms for the following variables: variance of earnings, variance of forecast errors, market equity/book equity ratio, and β . For each firm, the average level of the differencing variable was determined for the preswapping period and for the post-swapping period and the difference was calculated as: pre-mean - post-mean. Table 5.6 presents a test of the mean difference for each group against a null hypothesis of zero. For the difference in variance of earnings, the effective hedging treatment firms and the ineffective hedging control firms show a significant decrease, while the other groups have no significant change. A change in variance of earnings could signal a change in the predictability of earnings (i.e., a change in the level of earnings surprises) or it could signal a change in persistence (i.e., a change in the time series parameters of earnings). By examining the difference in the variance of forecast errors, the change in predictability can be tested. For all groups, there is no significant change in the variance of forecast errors, providing some evidence that the earnings were not easier to predict in the post-swapping period for any group of firms. Knowledge of this evidence is useful in drawing conclusions from the empirical tests of persistence and ERC changes.

Table 5.4
Univariate statistics on quarterly observations across all time periods

Panel A: Case 1 and	d 4 firms only: Eff	ective cash flow h	edging									
	Treatment fir	Treatment firms (56)					Control firms (50)					
	Median	Mean	Std. Dev.	Min.	Max.	Median	Mean	Std. Dev.	Min.	Max:		
Total assets	3315.2	5719.9	6496.6	101.5	36186.0	2146.4	4153.6	5922.7	101.6	33146.1		
Market equity	2371.4	4745.8	5654.5	23.4	38891.1	1613.8	4099.4	7534.1	25.0	74878.4		
Mk/Bk	1.664	2.107	1.613	.347	20.440	1.629	2.361	2.532	.039	44.052		
Debt/equity	.733	1.047	1.243	.0601	15.660	.656	1.213	2.140	.055	30.501		
β	1.174	1.193	.544	902	3.580	1.141	1.197	.588	836	3.670		
Firm quarters	1,976	1,976										
Panel B: Case 2 and	d 3 firms only: Inc	ffective cash flow	hedging									
	Treatment fi	rms (14)				Control firms (10)						
	Median	Mean	Std. Dev.	Min.	Max.	Median	Mean	Std. Dev.	Min.	Max.		
Total assets	4269.5	18056.5	22200.3	526.7	87707.0	2002.0	12494.4	16434.9	263.8	63859.0		
Market equity	4893.4	12110.8	16980.4	139.0	82120.6	3269.0	8997.8	10772.6	223.1	55952.7		
Mk/Bk	1.457	2.052	1.596	.429	11.074	1.627	2.348	2,447	.399	19.047		
Debt/equity	.790	1.167	.983	.056	6.393	.564	.669	.501	.018	2.449		
β	1.123	1.156	.583	165	2.753	1.028	1.113	.597	.056	3.316		
Firm quarters	494					309						

Table 5.5
Tests of means for selected variables between
Treatment and Control firms in each
time period (pre- and post-swapping)

Panel A: Case 1 and 4 firms only: Effective cash flow hedging							
	Treatment firms (56)		Control i	Control firms (50)		h _o :means are equal	
	Mean	Std. Dev.	Mean	Std. Dev.	t-stat.	prob>td	
Size -pre	4093	4360	3158	4702	1.0574	.2929	
Size -post	6135	6948	5575	9900	.3333	.7397	
Mk/Bk -pre	.684	.274	.851	.908	-1.2437	.2187	
M-/Bk -post	.6235	.361	.714	.571	9605	.3397	
β -pre	1.307	.382	1.317	.416	1350	.8929	
β -post	.921	.365	.944	.473	2701	.7877	
	<u> </u>		 		*		
Panel B: Case 2	2 and 3 firms	only: Ineffe	ective cash	flow hedgir	ng		
	Treatment	firms (14)	Control firms (10)		h _o :means are equal		
	Mean	Std. Dev.	Mean	Std. Dev.	t-stat.	prob>ltl	
Size -pre	9631	14656	10464	11661	1549	.8783	
Size -post	14527	20654	15039	17273	0659	.9480	
Mk/Bk -pre	.767	.321	.683	.289	.6690	.5109	
Mk/Bk -post	.582	.204	.597	.213	1863	.8542	
β-pre	1.35	.373	1.261	.326	.6211	.5412	
β -post	.874	.434	.806	.405	.3913	.6997	
			!		*		

Table 5.6 Tests of mean differences between pre- and post-swapping time periods for selected variables within Treatment and Control firm groups

	Treatment firms (56)			Control fi	Control firms (50)			
Differencing variable	Mean	Std. Dev.	t-stat.	prob>kl	Mean	Std. Dev.	t-stat.	prob>ltl
Variance of earnings	.085	.050	1.700	.0947	.174	.363	.478	.6344
Variance of forecast errors	-11.93	12.00	993	.3247	067	.225	302	.7639
Mk/Bk	.060	.034	1.764	.0832	.136	.116	1.175	.2454
β	.385	.039	9.643	.0001	.373	.049	7.581	.0001
Panel B: Case 2 and 3 firms	only: Ineffe	ctive cash	flow hedgir	ng				
	Treatmen	t firms (14)			Control firms (10)			
Differencing variable	Mean diff.	Std. Dev.	t-stat.	prob>kl	Mean	Std. Dev.	t-stat.	prob>ltl
Variance of earnings	.151	.107	.141	.1819	.091	.037	2.456	.0364
Variance of forecast errors	.060	.091	.654	.5244	-28.76	28.78	999	.3438
Mk/Bk	.185	.054	3.394	.0048	.085	.061	1.392	.1973
β	.476	.057	8.27	.0001	.454	.126	3.603	.0057

The significant decrease in β for all groups indicates a need to include a control for β in the empirical tests. Likewise, the significant decrease in the Market equity/Book equity ratio for both treatment groups indicates a need to control for this difference.

5.4.4 Statistics for eliminated firms

Tables 5.7, 5.8, and 5.9 provide descriptive statistics for the firms eliminated by the interest rate sensitivity analysis. These statistics are provided to demonstrate that there is no obvious characteristic that distinguishes these firms from the rest of the sample.

5.4.5 Industry representation

Table 5.10 presents information on the breakdown of industries represented in the sample. It is clear that the sample is not heavily dominated by one industry. Therefore, it is not expected that the results of the empirical tests will be driven by industry effects.

5.4.6 Economic significance of interest rate swaps

Table 5.11 presents descriptive information about the economic significance of the interest rate swaps for the effective cash flow hedging firms and for the ineffective cash flow hedging firms. Economic significance is measured with two ratios: the notional principal to long-term debt ratio and interest expense to sales ratio. Both measures are imperfect proxies for economic significance but both provide some descriptive level of swap activity. The notional principal to long-term debt ratio is imperfect because firms must disclose the total amount of notional principal and this total sometimes includes offsetting positions. For instance, one firm in the sample had a notional principal to long-term debt ratio of 16.84. Because this firm did not disclose the number of swaps or the

direction of individual swaps, it is not possible to determine how much of the long-term debt is covered on a net basis. Therefore, this ratio should only be used as a gauge of swap activity and can not really be meaningfully compared across the groups.

The interest expense to sales ratio is an attempt to determine whether or not changes to interest expense patterns would be significant enough to impact the valuation process. Because the interest expense related to swaps is not broken out separately in the income statement, the interest expense number may overstate the significance of the swap activity. This ratio does provide enough information to conclude that there is a slightly higher impact likely for the effective cash flow hedging firms (interest expense is approximately 3.5% of sales) than for the ineffective cash flow hedging firms (interest expense is approximately 2.9% of sales). This difference between groups is discussed in Chapter VIII as a possible limitation of the results.

Table 5.7 Univariate statistics on quarterly observations Across all time periods Firms eliminated after interest sensitivity analysis

	Treatment	firms (22)				Control fir	ms (21)			
	Median	Mean	Std. Dev.	Min.	Max.	Median	Mean	Std. Dev.	Min.	Max.
Total assets	2854.1	9613.5	28582.2	116.2	198938	2133.54	10933.1	32227.4	134.7	197279
Market equity	1579.3	3271.7	4472	149.1	32121	1568.1	5561.2	9172.4	89.8	64352.2
Bk/Mk	1.493	1.761	1.115	0.376	8.643	1.594	2.089	1.75	0.506	12.361
Debt/equity	0.858	1.641	2.232	0.105	14.221	0.612	0.922	1.15	0.057	8.688
β	1.161	1.203	0.568	-0.337	3.286	1.321	1.304	0.541	-0.015	2.972
Firm quarters	716					670				

Table 5.8

Tests of means for selected variables between eliminated Treatment and Control firms

	Treatmen	Treatment firms (22)		Control firms (21)		re equal
	Mean	Std.Dev.	Mean	Std.Dev.	t-stat.	prob>ltl
Size -pre	2792	3960	4262	6858	-0.8558	0.3985
Size -post	3782	4887	6273	11247	-0.9346	0.3583
Mk/Bk -pre	0.807	0.369	0.738	0.341	0.6368	0.5278
M-/Bk -post	0.735	0.394	0.606	0.302	1.209	0.2338
β-pre	1.349	0.324	1.421	0.421	-0.7095	0.482
β-post	0.931	0.461	1.063	0.418	-0.9969	0.3247

Table 5.9

Tests of mean differences between pre- and post-swapping time periods for selected variables within eliminated Treatment and Control firm groups

	Treatment firms (22)			Control firms (21)				
Differencing variable	Mean	Std.Dev.	t-stat.	prob>ltl	Mean	Std.Dev.	t-stat.	prob>ltl
Variance of earnings	0.013	0.1	0.137	0.8923	0.107	0.067	1.582	0.1294
Variance of forecast errors	0.041	0.045	0.909	0.3732	-0.01	0.018	-0.72	0.4807
Mk/Bk	0.071	0.05	1.412	0.0832	0.132	0.05	2.639	0.0157
β	0.419	0.071	5.946	0.0001	0.357	0.065	5.482	0.0001

Table 5.10
Industry breakdown for treatment and control firms, including Case 1 and 4 firms, Case 2 and 3 firms, and the eliminated firms

		Case 1 and 4		Case 2 and 3		Switche	rs
SIC#	Industry	treat	control	treat	control	treat	control
100-1999	Agricultural production,Oil and Gas extraction, Mining	5	4				
2000-2799	Food and paper products	10	9			7	5
2800-2999	Chemicals and allied products	6	6	6	5	5	4
3000-3999	Machinery, computer equipment, electronics, transportation equipment	19	16	6	3	5	8
4000-4999	water transportation, communication	4	5	1	ì		
4900-4999	utilities	6	5			3	2
5000-5999	wholesale and retail merchandising	3	3	1	1	2	2
7000-7999	hotels, business services, entertainment	3	2				
	Total firms	56	50	14	10	22	21

Table 5.11 Economic significance of interest rate swaps

	notional amount/ long-term debt	interest expense/ sales
Effective cash flow hedging firms	0.3935	0.0355
Ineffective cash flow hedging firms	0.282	0.0286

5.5 Summary

This chapter has provided information about the sample selection process and the categorization of firms for testing. In addition, descriptive statistics have been presented for a number of variables. The results of these analyses indicate that the selected control firms provide a successful match for the treatment firms and provide confidence that the empirical tests will not be adversely impacted by sample selection biases. The following chapter presents the empirical model and results for the test of the impact of interest rate swap activity on earnings persistence.

CHAPTER VI

EARNINGS PERSISTENCE: EMPIRICAL MODEL AND REGRESSION RESULTS

6.1 Introduction

This chapter presents the empirical model for testing the impact of implementing a derivative strategy on earnings persistence. Also presented are results of running the empirical model using the sample of firms described in Chapter V.

6.2 Earnings persistence model

As in prior research, this study uses analysts' forecast errors to proxy for the surprise in earnings. Analysts' forecast errors have been shown to be related to abnormal returns and analysts' forecasts have been shown to be an effective proxy for market expectations (Abarbanell 1991). This study uses the Value Line Investment Survey as the source of analyst forecasts.³⁹ The primary prediction examines the persistence of earnings, based upon a model from Easton and Zmijewski(1989). They use an analysts' revision model of the following form:

$$REV_{jt} = \theta_{j0} + \theta_{j1}FE_{jt} + \theta_{j2}PVLVL_{jt} + u_{jt},$$
 (13)

³⁹Value Line forecasts were deemed more appropriate in this context than were the consensus forecasts found on IBES. Revisions of forecasts are difficult to isolate and understand in a consensus format due to the various reasons analysts may have for updating. Forecasts from one source may be biased, but the bias is likely to be constant across treatment and control firms and should not impact the interpretation of the results.

where

 REV_{jt} = the revision in analyst forecast for period t+1 following the earnings announcement in quarter t.

FE_{it} = the forecast error for quarter t,

PVLVL_{jt} = the change in stock price for firm j from the earnings forecast date preceding the earnings announcement date through the earnings forecast date immediately after the earnings announcement date, excluding the three days before, the day of, and the three days after the earnings announcement date.

The coefficient, θ_{j1} , on the forecast error is the revision coefficient in Easton and Zmijewski's model. This coefficient captures all factors that influence a revision in the analyst's forecast beyond those captured by θ_{j2} , including but not limited to the information contained in the earnings announcement.⁴⁰ In the current study, the revision coefficient is used to proxy for the earnings persistence. Two slope indicator variables are added to the Easton and Zmijewski model: one to designate the post-hedging periods and one to allow for treatment and control firm differences in the pre-hedging period. The resulting empirical model is of the following form:

MODEL 1:

$$REVP_{jq} = \alpha_0 + \alpha_1 FEP_{jq} + \alpha_2 FEP_{jq} * Post_{jq} + \alpha_3 FEP_{jq} * Treat_j + \alpha_4 FEP_{jq} * Post_{jq} * Treat_j + \epsilon_{jq},$$

$$(14)$$

 $^{^{40}}$ The PVLVL_{jt} variable was weakly significant in Easton and Zmijewski's tests and is not used in the current model.

where

REVP_{jq} = The percentage revision in the quarter q+1 earnings forecast for firm j following the announcement of firm j quarter q earnings where revision is measured as the difference between the quarter q+1 forecast after announcement of quarter q earnings and the last quarter q+1 forecast preceding the announcement of quarter q earnings as reported in the Value Line Investment Survey.⁴¹

FEP_{jq} = The percentage forecast error for firm j for quarter q, measured as the difference between the actual earnings per share and the latest forecasted earnings per share for quarter q where forecasts and actual earnings per share were taken from the Value Line Investment Survey.

Post_{jq} = 1 if quarter q is in firm j's post-disclosure period and 0 otherwise.

Treat_i = 1 if firm j is a treatment firm and 0 otherwise.

Exhibit 6.1 shows how the coefficients in the model relate to the effect of interest rate swaps on earnings. In Exhibit 6.1, α_4 is the coefficient of primary interest. This coefficient shows the change in treatment firm persistence after controlling for the change in all firms over time and after controlling for any difference between treatment and control firms. The prediction made in the first part of this research is that α_4 will be positive if firms are hedging cash flows with interest rate swaps or α_4 will be zero or negative if firms' strategies are inconsistent with cash flow hedging.

⁴¹The percentage revision and percentage forecast error are used in this study to scale for size differences across firms. Easton and Zmijewski ran the model for individual firms and accordingly, used the nominal revisions and forecast errors.

Exhibit 6.1 Coefficients in Model 1

	Time period	
Type of Firm	Pre-swapping period	Post-swapping period
Control Firms	αι	$\alpha_1 + \alpha_2$
Treatment Firms	$\alpha_1 + \alpha_3$	$\alpha_1 + \alpha_2 + \alpha_3 + \alpha_4$

6.3 Test of persistence response

6.3.1 Regression results for Model 1

Table 6.1 provides the estimation results of running Model 1. Panel A presents the results of the model using all firms. Panels B and C present the results for the effective cash flow hedging firms (case 1 and 4 firms) and the ineffective cash flow hedging firms (case 2 and 3 firms), respectively. In Panel A, the revision coefficient which proxies for persistence increased for treatment firms in the period after disclosure after controlling for differences between treatment and control firms and controlling for general differences in the time periods, as evidenced by the significant positive coefficient on α_4 , the FEP post *Post* Post* Preat* variable.

As expected, the FEP_{jq} coefficient was significantly positive, establishing the base

⁴²The total number of firm quarters tested, 5752, is made up of the 3595 case 1 and 4 firm quarters, the 794 case 2 and 3 firm quarters, and the 1363 firm quarters for firms eliminated from the case categories.

⁴³Upon a review of influential observations, it was noted that approximately 10 quarterly observations were substantially affecting the coefficient estimates. Therefore, these extreme forecast error percentages and extreme revision percentages were winsorized to the values at the 1 and 99 percentile. This maintains the direction but dampens the relative magnitude of the effect for these extreme observations.

relation between the forecast errors and the forecast revisions. Panel A shows that the treatment firms had lower revision coefficients than the control firms in the pre-disclosure period, as evidenced by the negative coefficient on the treatment indicator variable, FEP_{ia}*Treat_i. In the post-disclosure period, panel A shows that all firms had lower revision coefficients as shown by the negative coefficient on the post-disclosure period indicator variable, FEP_{jq}*Post_{jq}. The significant positive coefficient on the postdisclosure treatment indicator, FEP_{iq}*Post_{iq}*Treat_i, is consistent with the prediction for effective cash hedging when all firms are included. Therefore, if no differentiation is made between firms' net asset position before swapping, it appears that firms, on average, effectively hedge cash flows with their interest rate swaps. However, in order to provide a stronger test of the predictions, Panel B presents the results of running Model 1 for the effective cash flow hedging firms (Cases 1 and 4) and panel C presents the results for the ineffective cash flow hedging firms (Cases 2 and 3). The coefficient on the postdisclosure period treatment indicator, FEP_{jq}*Post_{jq}*Treat_j, remains significantly positive in panel B, as predicted, and all other coefficients are consistent with Panel A. In panel C, however, the coefficient on FEP_{jq}*Post_{jq}*Treat_j is insignificant, while all other coefficients are consistent with panels A and B. The results in Table 6.1 are consistent with the prediction that an effective cash flow hedging strategy (as predicted for cases 1 and 4 in panel B) will increase persistence while a strategy that is inconsistent with cash flow hedging (as predicted for cases 2 and 3 in panel B) will not increase earnings persistence.

Table 6.1: Model 1 - Test of persistence response

$$REVP_{iq} = \alpha_{1}FEP_{iq} + \alpha_{2}FEP_{iq} * Post_{iq} + \alpha_{3}FEP_{iq} * Treat_{i}$$
$$+\alpha_{4}FEP_{iq} * Post_{iq} * Treat_{i} + \epsilon_{iq}$$

	Panel A: All	Firms	
	coefficient estimate	White's	two-tailed
variable		t-statistic	prob. > t
FEP _{jq}	0.23636	14.908	.0001
FEP _{jq} *Post _{jq}	-0.026371	-2.238	.025
FEP _{iq} *Treat _j	-0.010244	-1.855	.0636
FEP _{iq} *Post _{jq} *Treat _j	0.061558	3.341	.0008
# of observations	5752		
Adjusted R ²	.1947		
Panel B: Firm	ns categorized as cases 1 o	r 4: Effective cash flow he	dging
variable	coefficient estimates	White's t-statistic	one-tail prob.>t
FEP _{jq}	0.252519	12.98398	.0001
FEP _{iq} *Post _{jq}	-0.0359	-2.5589	.0054
FEP _{iq} *Treat _j	-0.01032	-1.9718	.0244
FEP _{jq} *Post _{jq} *Treat _j	0.086354	3.9264	.0001
# of observations	3595		
Adjusted R ²	.2444		
Panel C: Firm	s categorized as cases 2 or	3: Ineffective cash flow h	edging
FEP _{iq}	0.35523	5.5266	.0001
FEP _{iq} *Post _{jq}	-0.14472	-1.9325	.0268
FEP _{in} *Treat _i	-0.09166	-2.1221	.017
FEP _{jq} *Post _{jq} *Treat _j	0.052216	.6502	.2578
# of observations	794		
Adjusted R ²	0.1814		

6.3.2 Regression results for Model 1 with firm-specific and time-specific indicators

Any test of cross-sectionally pooled time-series data is subject to measurement error resulting from the forcing of firm and time specific influences into one intercept term. The appropriate solution for this problem depends upon the assumptions made about the firm and time specific intercept terms and can take the form of a fixed effects model or an error components model. A fixed effects model does not make any specific assumptions about the distribution of the firm and time specific terms and can accordingly be used for a wider range of problems than can an error components model.

Herein, the least restrictive assumption is made so that the firm and time specific influences are assumed to be fixed and a fixed effects model is then appropriate.

Therefore, Model 1 was augmented to incorporate an indicator variable for each firm and for each quarter, to control for both firm-specific and time-period specific influences.

The augmented model is as follows:

$$REVP_{jq} = \alpha_1 FEP_{jq} + \alpha_2 FEP_{jq} *Post_{jq} + \alpha_3 FEP_{jq} *Treat_j + \alpha_4 FEP_{jq} *Post_{jq} *Treat_j + \sum_{j=1}^{175} \alpha_{4-j} Dfirm_j + \sum_{q=1}^{44} \alpha_{179+j} Dtime_q + \epsilon_{jq}$$

$$(15)$$

where Dfirm is an indicator variable for each firm j, and

⁴⁴The ideas presented in this paragraph are based upon the discussion presented by Judge, Hill, Griffiths, Lutkepohl, and Lee (1982)in Chapter 16 of "Introduction to the Theory and Practice of Econometrics".

⁴⁵This methodology is applied in recent studies by Loudder, Boatsman, and Khurana (1996) and Abody (1996) where cross-sectional, time-series data is used.

Dtime is an indiator variable for each quarter t.

Table 6.2 presents the results of running Model 1 using all firms(panel A), firms in cases 1 and 4 (panel B) - the effective cash flow hedgers, and firms in cases 2 and 3 (panel C) - the ineffective cash flow hedgers. The table suppresses the coefficients on individual firms and time periods (significant coefficients found for: 19 firms/0 quarters in panel A, 14 firms/0 quarters in panel B, 4 firms/3 quarters in panel C). In panel A of table 6.2, the revision coefficient which proxies for persistence increased for treatment firms in the period after disclosure after controlling for differences between treatment and control firms and controlling for general differences in the time periods, as evidenced by the significant positive coefficient on α_4 , the FEP_{jq}*Post_{jq}*Treat_j variable.⁴⁶ The significant positive coefficient on FEP_{jq}*Post_{jq}*Treat_j is consistent with the prediction for effective cash hedging when all firms are included. Therefore, if no differentiation is made between firms' net asset position before swapping, it appears that firms, on average, effectively hedge cash flows with their interest rate swaps.

When firms' net asset position before swapping is considered, the revision coefficient increased for effective cash flow hedgers as evidenced by the significantly positive coefficient on the FEP_{jq}*Post_{jq}*Treat_j variable in panel B. For the ineffective cash flow hedgers shown in panel C, the coefficient on FEP_{jq}*Post_{jq}*Treat_j is positive but insignificant.

⁴⁶Once again, the extreme forecast error percentages and extreme revision percentages were winsorized to the 1 and 99 percentile. This procedure dampened the level of significance for all coefficients but had no impact on the direction of the coefficients in the model.

Table 6.2: Model 1 - Test of persistence response with indicator variables for each firm and each quarter

$$REVP_{jq} = \alpha_1 FEP_{jq} + \alpha_2 FEP_{jq} * Post_{jq} + \alpha_3 FEP_{jq} * Treat_j + \alpha_4 FEP_{jq} * Post_{jq} * Treat_j$$

$$+ \sum_{j=1}^{175} \alpha_4 \cdot Dfirm_j + \sum_{q=1}^{44} \alpha_{179} \cdot Dtime_q + \epsilon_{jq}$$

Panel A: All Firms							
variable	coefficient estimate	White's t-statistic	two-tail prob. > t				
FEP _{jq}	.223193	14.404	.0001				
FEP _{iq} *Post _{iq}	-0.030214	-2.677	.0074				
FEP _{iq} *Treat _j	-0.006906	-1.275	.2023				
FEP _{jq} *Post _{jq} *Treat _j	0.060472	3.584	.0004				
# of observations	5752						
Adjusted R ²	.2775						
Panel B: F	irms categorized as cases 1	or 4: Effective cash flow	v hedging				
variable	coefficient estimates	White's t-statistic	one-tail prob.>t				
FEP _{iq}	0.245336	12.9648	.0001				
FEP _{jq} *Post _{jq}	-0.040287	-2.9585	.0018				
FEP _{jq} *Treat _j	-0.008142	-1.4535	.0695				
FEP _{jq} *Post _{jq} *Treat _j	0.084184	4.1655	.0001				
# of observations	3595						
Adjusted R ²	.3109						
Panel C: Fi	rms categorized as cases 2	or 3: Ineffective cash flo	w hedging				
FEP _{jq}	0.322290	5.0253	.0001				
FEP _{jq} *Post _{jq}	-0.120492	-1.7301	.0416				
FEP _{jq} *Treat _j	-0.063156	-1.8335	.0325				
FEP _{jq} *Post _{jq} *Treat _j	0.029368	.3992	.3474				
# of observations	794						
Adjusted R ²	.2853						

As expected, the FEP_{jq} coefficient was significantly positive in all panels, establishing the base relation between the forecast errors and the forecast revisions. Table 6.2 shows that there was no significant difference in panels A and B between the treatment firms and the control firms in the pre-disclosure period, as evidenced by the insignificant coefficient on the treatment indicator variable, FEP_{jq}*Treat_j, while the Panel C treatment firms had significantly lower revision coefficients in the pre-disclosure period. In the post-disclosure period, each panel of table 4 shows that all firms had lower revision coefficients as shown by the significantly negative coefficient on the post-disclosure period indicator variable, FEP_{jq}*Post_{jq}. Again, these collective results are consistent with the predictions for effective cash flow hedging and ineffective cash flow hedging. The inclusion of the firm and time specific indicator variables did not change the interpretation of the model's results.

6.4 Summary

In summary, the tests performed in this chapter support the predictions presented in Chapter 4 for the firm-specific impact of effective cash flow hedging and ineffective cash flow hedging on the persistence of earnings. For this study, persistence is proxied by the relation between revisions of analysts forecasts of future earnings and current earnings surprise, described as a revision coefficient. An increase in the revision coefficient in the period following disclosure of interest rate swap activity is interpreted as an increase in the persistence of the earnings. The findings support the prediction that effective cash flow hedging of interest rates will increase the persistence of earnings while ineffective cash flow hedging will not.

CHAPTER VII

EARNINGS RESPONSE COEFFICIENT EMPIRICAL MODEL AND REGRESSION RESULTS

7.1 Introduction

This chapter presents the empirical model for testing the impact of implementing an interest rate hedging strategy on the earnings response coefficient. Also presented are the results of running the empirical model using the sample of firms described in Chapter V.

7.2 The earnings response coefficient model

The second prediction made in this study addresses the change in ERC following the implementation of the hedging strategy. This prediction is tested using a model similar to that found in Collins and Salatka (1993). The empirical model is as follows:

MODEL 2

$$AR_{jq}/p_{jt-2} = \beta_{1}FE_{jq}/p_{jt-2} + \beta_{2}FE_{jq} * Post_{jq}/p_{jt-2} + \beta_{3}FE_{jq} * Treat_{j}/p_{jt-2} + \beta_{4}FEP_{jq} * Post_{jq} * Treat_{j}/p_{jt-2} + \beta_{5}FE_{jq} * EBETA_{jq}/p_{jt-2} + \beta_{6}FE_{jq} * MK/BK_{jq}/p_{jt-2} + \sum_{j=1}^{175} \beta_{6 - j}Dfirm_{j}/p_{jt-2} + \sum_{q=1}^{44} \beta_{181 - j}Dtime_{q}/p_{jt-2} + \epsilon_{jq}$$
(16)

where AR_{jq} = The abnormal return, measured as the prediction error from a market model run over day t-1 and day t where t is the quarterly earnings announcement date for quarter q. The parameters of the market model are estimated using a 200 day trading interval from

- day t-100 to day $t+100^{47}$,
- Fe_{jq} = The nominal forecast error where forecast error is as defined in equation 10 above. This specification follows that of Collins and Salatka(1993),
- Beta_{jq} = The market model beta for firm j for quarter q where the parameters of the market model are estimated using a 200 day trading interval from day t-100 to day t+100.
- MK/BK $_{jq}$ = The market equity to book equity ratio for firm j for quarter q where market equity and book equity are measured at the end of the fiscal quarter,
- p_{jt-2} = The price of firm j stock two days prior to the quarterly earnings announcement date, and all other variables are as described previously.

Based upon the findings of prior research regarding cross-sectional determinants of ERCs, the model includes a proxy for risk (Beta_{jq}) and a proxy for growth (Mk/Bk_{jq}). These are included as control variables and are expected to have the signs found in prior research, where ERCs are found to be associated negatively with risk and positively with growth. The model does not include a control variable for interest rate levels, as this intertemporal ERC determinant is controlled via the control sample. Exhibit 7.1 summarizes the coefficient combinations necessary to evaluate the ERC's of the treatment

⁴⁷In estimating the market model parameters, daily returns were examined for extreme market return days and these were eliminated from the estimation model. Excluded due to extreme returns were most of the days in October 1987 subsequent to the 15th as well as days in other years where the market return for the day was more than 5 standard deviations away from the average return for the estimation period. These days were also eliminated as abnormal return dates, resulting in the loss of several firm quarter observations and these dates were eliminated as scaling days for the regression model. Bowen, Johnson, and Shevlin (1989) examine the return-earnings relation during the market crash of October 1987 and find that the relation is maintained in their tests except for the days Oct. 19-20. However, the inclusion of these extreme return observations distorts the estimated coefficients in a pooled cross-sectional setting such as the current study.

and control samples in the pre- and post-swapping period.

Exhibit 7.1
Coefficients in Model 2

	Time period	
Type of Firm	Pre-swapping period	Post-swapping period
Control Firms	β,	$\beta_1 + \beta_2$
Treatment Firms	$\beta_1 + \beta_3$	$\beta_1 + \beta_2 + \beta_3 + \beta_4$

The coefficient of interest for the tests of the second model is β_4 . Analogous to the persistence model, this coefficient captures the change in ERC after implementation of the derivative strategy beyond any general time trend and controlling for differences between the treatment and control firms. The prediction is that β_4 will be positive if a cash flow hedging strategy is used and that β_4 will be zero or negative if the strategy is inconsistent with cash flow hedging.

7.3 Test of ERC response

7.3.1 Regression results for Model 2

Turning to tests of the ERC response to firms' hedging strategies, Table 7.1 presents the results of running Model 2 for all firms (panel A), for the effective cash flow hedging firms (panel B), and for the ineffective cash flow hedging firms (panel C). Each panel is discussed below.

Panel A is presented as a base result that does not control for differences in the effectiveness of the hedging strategy. The signs on the control variables of market beta (-) and market/book ratio (+) were consistent with prior research although neither

variable was significantly different from zero. As expected, the FE_{jq} coefficient was significantly positive, establishing the base relation between unexpected earnings and abnormal returns. In table 7.1 and consistent with the findings in the persistence tests, the coefficient on $Fe_{jq}*Post_{jq}$ was significantly negative, indicating that ERCs in the post-disclosure period were lower across all firms in the sample. The treatment indicator, $FE_{jq}*Treat_{j}$ was not significantly different from zero, indicating that the treatment firms and control firms were not significantly different in the pre-disclosure period. The insignificantly positive $FE_{jq}*Post_{jq}*Treat_{j}$ coefficient indicates that for the sample of all firms, there is no significant impact on the ERC from the implementation of the hedging strategy. However, a more powerful test of the predictions can be performed by partitioning the data according to the expected effectiveness of the hedging strategies as described in Chapter V.

Panel B provides results from running Model 2 for the combination of cases 1 and 4 - the effective cash flow hedging firms. The results in panel B are consistent with those in panel A, with one interesting exception. For this group of effective cash flow hedging firms, the coefficient on the hedging strategy effect, $FE_{jq}*Post_{jq}*Treat_j$, is significantly positive. This finding is consistent with the prediction for an effective hedging strategy.

The results in panel C are interesting as well. All coefficients are consistent with those in panel A except that the $FE_{jq}^*Post_{jq}$ coefficient is no longer significant, indicating that for this group of ineffective cash flow hedging firms there was no systematic decline in ERCs across time. The insignificant coefficient on the hedging strategy effect,

Table 7.1: Model 2 - Test of the ERC response

$$AR_{,q}/p_{,p-2} = \beta_{1}FE_{,q}/p_{,p-2} + \beta_{2}FE_{,q}*Post_{,q}/p_{,p-2} + \beta_{3}FE_{,q}*Treat_{,p}/p_{,p-2} + \beta_{4}FE_{,q}*Post_{,q}*Treat_{,p}/p_{,p-2} + \beta_{5}FE_{,q}*EBETA_{,q}/p_{,p-2} + \beta_{6}FE_{,q}*MK/BK_{,q}/p_{,p-2} + + \epsilon_{,q}$$

Panel A: All Firms							
variable	coefficient estimate	White's t-statistic	two-tailed prob.> t				
FE _{jq}	0.810421	6.945	.0001				
FE _{jq} *Post _{jq}	-0.143757	-2.947	.0032				
FE _{jq} *Treat _j	-0.026194	-0.223	.8258				
FE _{jq} *Post _{jq} *Treat _j	0.183598	1.040	.2984				
FE _{jq} *Beta _{jq}	-0.02029	-1.477	.1388				
FE _{jq} *MK/BK _{jq}	0.048549	1.376	.1691				
# of observations	5851						
Adjusted R ²	.0381						
Panel B: Firm	s categorized as cases 1	or 4: Effective cash	flow hedging				
variable	coefficient estimate	White's t-statistic	one-tailed prob. >t				
FE _{jq}	0.661361	4.456	.0001				
FE _{jq} *Post _{jq}	-0.1397	-2.602	.0047				
FE _{jq} *Treat _j	-0.12852	-1.092	.1379				
Fe _{jq} *Post _{jq} *Treat _j	0.271769	1.431	.0764				
FE _{jq} *Beta _{jq}	-0.01919	-1.420	.0778				
FE _{jq} *Mk/Bk _{jq}	0.07296	1.0117	.1562				
# of observations	3673						
Adjusted R	.0256						

Panel C: Firms categorized as cases 2 or 3:Ineffective cash flow hedging				
variable	coefficient	White's	One-tailed	
	estimate	t-statistic	prob. > t	
FE _{jq}	0.833709	1.073	.1423	
FE _{jq} *Post _{jq}	-0.79789	-1.246	.1056	
FE _{jq} *Treat _j	02577	05232	.4801	
FE _{jq} *Post _{jq} *Treat _{i4}	1.116835	1.249	.1056	
FE _{jq} *Beta _{jq}	05429	1513	.4404	
FE _{jq} *Mk/Bk _{jq}	0.1102	.3372	.3669	
# of observations	803			
Adjusted R ²	.0305			

Note:

 AR_{jq} is the abnormal return, measured as the prediction error from a market model run over day t-1 and day t where t is the quarterly earnings announcement date for quarter q.

FE_{iq} is the forecast error for quarter q for firm j,

Postin is an indicator variable taking the value of 1 for quarters after disclosure of swap and 0 otherwise,

Treat, is an indicator variable taking the value of 1 for treatment firms and 0 for control firms,

Beta, is the market model beta for firm j estimated using a 200 day period around t.

MK/BK_{iq} is the market equity/book equity for firm j for quarter q,

DFirm is an indicator variable for each firm,

Dtime is an indicator variable for each quarter,

and

 p_{it-2} is the price for firm j at day t-2 and is used as a scalar in the above model.

FE_{jq}*Post _{jq}*Treat_j, can be interpreted as consistent with the predictions for an ineffective hedging strategy.

7.3.2 Regression results for Model 2 with firm-specific and time-specific indicators

Finally, similar to the method described in chapter VI, Model 2 was run as a fixed effects model, with indicator variables added for each firm and for each quarter. The indicator variables control for firm-specific and time period-specific omitted variables that weaken the power of the simpler test presented in Table 7.1. The augmented model is shown below:

$$AR_{jq}/p_{jt-2} = \beta_{1}FE_{jq}/p_{jt-2} + \beta_{2}FE_{jq} * Post_{jq}/p_{jt-2} + \beta_{3}FE_{jq} * Treat_{j}/p_{jt-2} + \beta_{4}FEP_{jq} * Post_{jq} * Treat_{j}/p_{jt-2} + \beta_{5}FE_{jq} * EBETA_{jq}/p_{jt-2} + \beta_{6}FE_{jq} * MK/BK_{jq}/p_{jt-2} + \sum_{j=1}^{175} \beta_{6+j}Dfirm_{j}/p_{jt-2} + \sum_{q=1}^{44} \beta_{181+j}Dtime_{q}/p_{jt-2} + \epsilon_{jq}$$
(17)

where Dfirm is an indicator variable for each firm and Dtime is an indicator variable for each quarter.

Table 7.2 presents the results from this model for all firms (panel A), case 1 and 4 firms (panel B), case 2 and 3 firms (panel C). The coefficients on the indicator variables are suppressed (significant coefficient found for: 2 firms/0 quarters in panel A, 10 firms/4 quarters in panel B, 1 firm/5 quarters in panel C). Results across all panels are consistent with those in Table 7.1. The hedging strategy impact variable, $FE_{jq}*Post_{jq}*Treat_{j'}$ is positively significant in panel B (consistent with the categorization of these firms as "effective cash flow hedgers") and is negative in panel C (the firms

categorized as "ineffective cash flow hedgers"). Therefore, the results are considered to be consistent with the predictions for effective versus ineffective cash flow hedging of interest rate risk. The firms that were categorized as effective cash flow hedging firms experienced an increase in ERCs following the implementation of the interest rate swap strategy while those firms that were categorized as ineffective cash flow hedging firms did not experience a change in ERC corresponding to the implementation of the interest rate swap strategy.

7.4 Summary

This chapter presented the empirical model and results for the tests of the ERC response to implementation of an interest rate swap strategy. The results, based upon a model augmented to include firm-specific and time period-specific indicator variables, support the predictions made in Chapter IV. The implementation of the interest rate swap strategy was associated with an increase in the ERCs of firms categorized as effective cash flow hedgers. For firms categorized as ineffective cash flow hedgers, there was no significant change in the ERCs. The following chapter concludes this dissertation and offers ideas for future research based upon the findings and conclusions herein.

Table 7.2: Model 2 - Test of the ERC response with indicator variables for each firm and each quarter

$$AR_{,q}/p_{,t-2} = \beta_{1}FE_{,q}/p_{,t-2} + \beta_{2}FE_{,q} * Post_{,q}/p_{,t-2} + \beta_{3}FE_{,q} * Treat_{,p}/p_{,t-2} + \beta_{4}FE_{,q} * Post_{,q}/p_{,t-2} + \beta_{5}FE_{,q} * EBETA_{,q}/p_{,t-2} + \beta_{5}FE_{,q} * MK/BK_{,q}/p_{,t-2} + \sum_{j=1}^{175} \beta_{6,j}Dfirm_{,p}/p_{,t-2} + \sum_{q=1}^{44} \beta_{121,j}Dtime_{,q}/p_{,t-2} + \epsilon_{,q}$$

Panel A: All Firms					
variable	coefficient estimate	White's t-statistic	two-tailed prob. > t		
FE _{jq}	0.821060	7.189	.0001		
FE _{jq} *Post _{jq}	-0.114238	-1.846	.0651		
FE _{jq} *Treat _j	0.001578	.0119	.9840		
FE _{jq} *Post _{jq} *Treat _j	0.254534	1.375	.2023		
FE _{jq} *Beta _{jq}	-0.020873	-1.726	.0845		
FE _{jq} *MK/BK _{jq}	0.031368	.8587	.3926		
# of observations	5851				
Adjusted R ²	.0491				
Panel B: Firms categorized as cases 1 or 4: Effective cash flow hedging					
variable	coefficient estimate	White's t-statistic	one-tailed prob. >t		
FE _{jq}	0.683708	4.9151	.0001		
FE _{jq} *Post _{jq}	-0.122228	-1.7369	.0400		
FE _{jq} *Treat _j	-0.136286	-1.0568	.1444		
FE _{jq} *Post _{jq} *Treat _j	0.415752	2.1415	.0175		
FE _{jq} *Beta _{jq}	-0.020593	-1.778	.0375		
FE _{jq} *Mk/Bk _{jq}	0.053507	.715	.2347		
# of observations	3673				
Adjusted R	.0398				

Panel C: Firms categorized as cases 2 or 3:Ineffective cash flow hedging				
variable	coefficient estimate	White's t-statistic	One-tailed prob. > t	
FE _{jq}	0.832746	1.004	.1587	
FE_{jq} *Post $_{jq}$	-0.433499	6791	.2505	
$FE_{jq}^{*}Treat_{j}$	-0.378418	7593	.2269	
FE _{jq} *Post _{jq} *Treat _{j4}	1.06539	1.2631	.1025	
FE _{jq} *Beta _{jq}	-0.033405	09527	.4645	
FE _{jq} *Mk/Bk _{jq}	0.146835	.4347	.3328	
# of observations	803			
Adjusted R ²	.0503			

Note:

 AR_{jq} is the abnormal return, measured as the prediction error from a market model run over day t-1 and day t where t is the quarterly earnings announcement date for quarter q.

FEig is the forecast error for quarter q for firm j,

Post_{jq} is an indicator variable taking the value of 1 for quarters after disclosure of swap and 0 otherwise,

Treat, is an indicator variable taking the value of 1 for treatment firms and 0 for control firms,

Beta_{jq} is the market model beta for firm j estimated using a 200 day period around t.

MK/BK_{iq} is the market equity/book equity for firm j for quarter q,

DFirm is an indicator variable for each firm,

Dtime is an indicator variable for each quarter,

and

 p_{jt-2} is the price for firm j at day t-2 and is used as a scalar in the above model.

CHAPTER VIII

CONCLUSION AND SUGGESTIONS FOR FUTURE RESULTS

8.1 Interpretation of results

The results of the empirical tests in this dissertation are generally consistent with the predictions concerning the valuation relevance of interest rate swap strategies. Firms that (1) disclose the use of interest rate swaps and (2) swap to hedge based on net asset classification demonstrate changes in persistence and ERCs consistent with effective cash flow hedging. However, firms that (1) disclose the use of interest rate swaps as a tool to manage interest rates but (2) swap to a position inconsistent with hedging based on net asset classification demonstrate changes in persistence and ERCs that are inconsistent with effective cash flow hedging.

What do these results mean for regulators and standard setters? One could interpret these findings as evidence that even though derivative strategies are not transparent within the financial statements, the financial statement users are apparently able to correctly infer the valuation relevance of the strategies. The results for the group of "effective cash flow hedge" firms are consistent with the predictions even though the financial statements do not present the "effective cash flow hedge" positions explicitly. The lack of a significant result for firms in the "ineffective cash flow hedge" group is consistent with the findings by Collins and Salatka (1993) concerning the ERC/reporting quality relation. Firms in this group disclosed an interest rate swap as an interest rate

risk management tool but the direction of the swap, given the interest rate sensitivity of the firm, was inconsistent with an effective cash flow hedge, and the ERCs did not change in response to the strategy. 48

8.2 Limitations and suggestions for future research

One potential limitation of these results is the size difference between the firms effective cash flow hedging firms and ineffective cash flow hedging firms. As discussed in Chapter V, the ineffective cash flow hedging firms were larger on average than the effective cash flow hedging firms as shown in Table 5.4. Therefore, it is possible that the lack of significance for the swap activity is because for the larger firms the swap activity is not economically significant enough to impact the persistence or ERC. Table 5.11 also indicates that the interest expense to sales ratio is smaller for the ineffective cash flow hedging firms, another possible indicator of economic significance. Future research will need to address the size differential to rule out this limitation.

Another potentially limiting factor is the self-selection issue for the treatment firms. While attempts have been made to control for a variety of firm characteristics, there could still be omitted variables that are not captured in this dissertation.

This dissertation makes predictions about changes in persistence and ERCs. It

⁴⁸Some interesting implications of these findings concerning the most recently proposed accounting guidelines are as follows:

⁽¹⁾ firms may be hedging cash flows effectively through case 4 but this case would be reported as a fair value hedge under the latest FAS exposure draft, without cash flow effect transparency, and

⁽²⁾ a case 2 strategy would be accounted for as a cash flow hedge but firm-wide cash flow exposure would not be decreased by this strategy.

These contradictory disclosures underscore the board's desire to eventually require firm-level assessment of hedging strategies as stated in paragraphs 161-164 of the FAS exposure draft dated June 20, 1996.

does not address changes in predictability, a concept discussed by Lipe (1990). The data gathered for this dissertation could also be used in an examination of predictability of earnings following implementation of a derivative strategy.

The findings suggest that the interest rate management strategies taken by firms are valuation relevant. The results add empirical evidence to the mostly theoretical-based literature on the role and impact of hedging at the firm level. An interesting empirical question for future research will be to see how the persistence and ERC impacts are affected by changes in the reporting requirements.

The sample used in this study can also be used in future research of finance related value relevance hypotheses. For example, the investment protection hypothesis implies that market-to-book ratios should change for firms that successfully hedge. Data gathered for tests in this dissertation could easily be used to test this hypothesis.

Finally, the evidence provided by this study also adds additional support to the earlier work of Hentschel and Kothari(1995) who concluded that firms were not gambling on average with their derivative strategies. This study makes directional predictions for the impact on persistence and ERCs of effectively hedging cash flows and finds support for an average hedging strategy, thus strengthening the position that on average, firms are not speculating on cash flows with their interest rate swaps.

APPENDIX A

Following the method of decomposing a sequence presented by Enders (1996) using the methodology of Beveridge and Nelson (1981), the stochastic trend of the interest-rate sensitive series, C_t, is found below.

First, stating the model at C₁:

$$C_t = \gamma_t + (D+K)(\alpha + (1-\beta)R_{t-1} + \epsilon_t)$$
 or

$$C_t = \gamma_t + (D+K)\alpha + (D+K)(1-\beta)R_{t-1} + (D+K)\epsilon_t$$

where ϵ is the realization from a random process with $E(\epsilon_t) = 0$ and

$$Var(\epsilon_t) = \sigma^2_{\epsilon}$$

and recognizing $\gamma_t + (D+K)\alpha$ as a permanent deterministic drift in C_t , the stochastic trend of the series is derived by taking the expectation at time t of future C levels:

$$E_tC_{t+1} = (1-\beta)C_t,$$

$$E_tC_{t+2} = (1-\beta)(1-\beta)C_t$$

$$E_tC_{t+3} = (1-\beta)(1-\beta)(1-\beta)C_t$$

$$E_tC_{t+j}=(1-\beta)^jC_t.$$

Now, restating the last expression in terms of the surprise at period t, $(D+K)\in_{t}$, the permaneence factor, Ψ_{c} , for the series can be summarized as:

$$\Psi_c = (1 - \beta)^{j} (D + K) \in t$$
, for all periods t+j.

Again, following the method presented by Enders (1996) using the methodology of Beveridge and Nelson (1981), the stochastic trend of the seasonal series, S, is found below.

First, stating the model at S₁:

$$S_1 = S_{1.4} + \phi(S_{1.1} - S_{1.5}) + \theta_4 a_{1.4} + a_{1.5}$$

where a, is the realization at time t from a random process with

$$E(a) = 0$$
 and

$$Var(a) = \sigma^2$$

and taking the expectation at time t of future S levels, the stochastic trend of the series is derived as follows:

$$\begin{split} E_{t}S_{t+1} &= S_{t-3} + \varphi(S_{t} - S_{t-4}) + \theta_{4}a_{t-3} \\ E_{t}S_{t+2} &= S_{t-2} + \varphi((S_{t-3} + \varphi(S_{t} - S_{t-4}) + \theta_{4}a_{t-3}) - S_{t-3}) + \theta_{4}a_{t-2} \\ &= S_{t-2} + \varphi(\varphi(S_{t} - S_{t-4}) + \theta_{4}a_{t-3}) + \theta_{4}a_{t-2} \\ &= S_{t-2} + \varphi^{2}(S_{t} - S_{t-4}) + \varphi\theta_{4}a_{t-3} + \theta_{4}a_{t-2} \\ E_{t}S_{t+3} &= S_{t-1} + \varphi(S_{t-2} + \varphi(\varphi(S_{t} - S_{t-4}) + \theta_{4}a_{t-3}) + \theta_{4}a_{t-2} - S_{t-2}) + \theta_{4}a_{t-1} \end{split}$$

$$\begin{split} &= S_{t-1} + \varphi(\varphi(\varphi(S_t - S_{t-4}) + \theta_4 a_{t-3}) + \theta_4 a_{t-2}) + \theta_4 a_{t-1} \\ &= S_{t-1} + \varphi^3(S_t - S_{t-4}) + \varphi^2 \theta_4 a_{t-3} + \varphi \theta_4 a_{t-2} + \theta_4 a_{t-1} \,. \end{split}$$

Expressed in terms of the surprise at period t, a_t , the permanence of the shock for j periods into the future is found in the term that carries S_t into the future:

$$\Psi_s = \Phi^j a_t$$
 for j<4,

and for periods t+4 through t+7, the expectation at t is:

$$\begin{split} E_t S_{t+4} &= S_t + \varphi(\varphi(\cdot \varphi(\varphi(S_t - S_{t+4}) + \theta_4 a_{t+3}) + \theta_4 a_{t+2}) + \theta_4 a_{t+1}) + \theta_4 a_t \\ &= (1 + \varphi^4) S_t - \varphi^4 S_{t+4} + \varphi^3 \theta_4 a_{t+3} + \varphi^2 \theta_4 a_{t+2} + \varphi \theta_4 a_{t+1} + \theta_4 a_t \\ &= (1 + \varphi^4) S_t - \varphi^4 S_{t+4} + \varphi^3 \theta_4 a_{t+3} + \varphi^2 \theta_4 a_{t+2} + \varphi \theta_4 a_{t+1} + \theta_4 a_t \\ E_t S_{t+5} &= S_{t+3} + \varphi(S_t - S_{t+4}) + \theta_4 a_{t+3} + \varphi^5 (S_t - S_{t+4}) + \varphi^4 \theta_4 a_{t+1}) + \theta_4 a_t - S_t) \\ &= S_{t+3} + \varphi(S_t - S_{t+4}) + \theta_4 a_{t+3} + \varphi^5 (S_t - S_{t+4}) + \varphi^4 \theta_4 a_{t+3} + \varphi^3 \theta_4 a_{t+2} \\ &+ \varphi^2 \theta_4 a_{t+1} + \varphi \theta_4 a_t \\ E_t S_{t+6} &= S_{t+2} + \varphi(\varphi(S_t - S_{t+4}) + \theta_4 a_{t+3}) + \theta_4 a_{t+2} + \varphi(\varphi^5 (S_t - S_{t+4}) + \varphi^5 \theta_4 a_{t+3} \\ &+ \varphi^4 \theta_4 a_{t+3} + \varphi^3 \theta_4 a_{t+2} + \varphi^2 \theta_4 a_{t+1} + \varphi \theta_4 a_t \\ &= S_{t+2} + \varphi^2 (S_t - S_{t+4}) + \varphi \theta_4 a_{t+3} + \theta_4 a_{t+2} + \varphi^6 (S_t - S_{t+4}) + \varphi^5 \theta_4 a_{t+3} \\ &+ \varphi^4 \theta_4 a_{t+2} + \varphi^3 \theta_4 a_{t+1} + \varphi^2 \theta_4 a_t \\ E_t S_{t+7} &= S_{t+1} + \varphi(\varphi(\varphi(S_t - S_{t+4}) + \theta_4 a_{t+3}) + \theta_4 a_{t+2} + \varphi^3 \theta_4 a_{t+1} + \varphi^2 \theta_4 a_t) \end{split}$$

$$= S_{t-1} + \phi^{3}(S_{t} - S_{t-4}) + \phi^{2} \theta_{4} a_{t-3} + \phi \theta_{4} a_{t-2} + \phi^{7}(S_{t} - S_{t-4})$$
$$+ \phi^{6} \theta_{4} a_{t-3} + \phi^{5} \theta_{4} a_{t-2} + \phi^{4} \theta_{4} a_{t-1} + \phi^{3} \theta_{4} a_{t}$$

which in terms of the surprise at period t, a_t , (found by gathering terms for S_t and a_t) is as follows for these four periods:

$$\Psi_s = \phi^{j-4} a_t + \phi^j a_t + \phi^{j-4} \theta_4 a_t \qquad \text{for } 4 \le j < 8,$$

and the process continues likewise for periods beyond j+8.

REFERENCES

- Abarbanell, Jeffrey (1991). "Do Analysts' Earnings Forecasts Incorporate Information In Prior Stock Price Changes?" <u>Journal of Accounting and Economics</u> vol. 14, pp.147-165.
- Arak, M., A.Estrella, L. Goodman, and A. Silver (1988). "Interest Rate Swaps: An Alternative Explanation," Financial Management, 12-18.
- Asquith, P. and D.W. Mullins (1983). "Equity issues and Offering Dilution," <u>Journal of Financial Economics</u> vol. 15, pp. 61-89.
- Beaver, W., R. Lambert and S. Ryan (1987). "The information content of security prices: A second look," <u>Journal of Accounting and Economics</u> vol. 9, pp. 139-158.
- Bernard, V. and J. Thomas (1990). "Evidence that Stock Prices Do Not Fully Reflect the Implications of Current Earnings for Future Earnings," <u>Journal of Accounting and Economics</u> vol 13, pp. 305-340.
- Bowen, R., M. Johnson, and T. Shevlin (1989). "Informational Efficiency and the Information Content of Earnings during the Market Crash of October 1987," <u>Journal of Accounting and Economics</u> vol. 11, pp. 225-254.
- Brenner, R. J., R.H. Harjes, and K.F. Kroner (1996). "Another Look at Models of the Short-Term Interest Rate," <u>Journal of Financial and Quantitative Analysis</u> vol. 31, pp.85-107
- Brown, L. D., and M.S. Rozeff (1979). "Univariate Time-Series Models of Quarterly Accounting Earnings per Share: A Proposed Model," <u>Journal of Accounting Research</u> vol. 17, pp. 179-189.
- Chan, K.C, G. Karolyi, Andrew, F. Longstaff, A. Sanders (1992). "An Empirical Comparison of Alternative Models of the Short-Term Interest Rate," <u>The Journal of Finance</u> July, pp. 1209-1227.
- Collins, D.W., and S. P. Kothari (1989). "An Analysis of the Intertemporal and Cross-Sectional Determinants of Earnings Response Coefficients," <u>Journal of Accounting and Economics</u> July, pp.141-181.
- Collins, D.W., and W. Salatka (1988). "Noisy Accounting Earnings Signals and Earnings Response Coefficients: The Case of Foreign Currency Accounting,"

 <u>ContemporaryAccounting Research</u> Fall, pp.119-159.

- Cox, John C., J. Ingersoll, Jr., and S. Ross (1980). "An Analysis of Variable Rate Loan Contracts," The Journal of Finance vol. 35(2), pp. 389-403.
- Dann, L.Y., and W.H. Mikkelson (1984). "Convertible debt issuance, capital structure change and financing-related information: Some new evidence," <u>Journal of Financial Economics</u> vol. 13, pp.157-186.
- Demarzo, P.M., and D. Duffie (1995). "Corporate Incentives for Hedging and Hedgd Accounting," The Review of Financial Studies, vol 8, pp. 743-771.
- Easton, P.D., and M.E. Zmijewske (1989). "Cross-Sectional Variation in the Stock Market Response to the Announcement of Accounting Earnings," <u>Journal of Accounting and Economics</u> July, pp.117-141.
- Enders, Walter (1995). Applied Econometric Time Series. (Wiley, New York, NY).
- Fenn, G., M. Post, and S.A. Sharpe (1996a). "Debt maturity and the Use of Interest Rate Derivatives by Nonfinancial Firms," FEDS Working Paper 96-36.
- Fenn, G., M. Post, and S.A. Sharpe (1996). "Does Corporate Risk Management Create Shareholder Value? A Survey of Economic Theory and Evidence," working paper, Capital Markets Section, Federal Reserve Board, forthcoming in <u>Financial Risk and the Corporate Treasury</u>, Risk Publications, London.
- Financial Accounting Series, (1996), FASB, February 26.
- Financial Accounting Standards Board (1996) "Accounting for Derivative and Similar Financial Instruments and for Hedging Activities," exposure draft, June 20.
- Flavin, M. (1981). "The adjustment of consumption to changing expectations about future income," <u>Journal of Political Economy</u> vol. 89, pp.974-1009.
- Froot, K.A., D. Scharfstein, and J.C. Stein (1993)"A Framework for Risk Management,"

 <u>Journal of Applied Corporate Finance</u> Fall, pp.22-32.
- Froot, K.A., D. Scharfstein, and J.C.Stein (1993) "Risk Management:Coordinating Corporate Investment and Financing Policies," <u>The Journal of Finance</u> vol. 48, pp.1629-1658.
- GAO Report on Financial Derivatives, 1994.
- Geczy, C., B. Minton, and C. Schrand (1995). "Why Firms Use Derivatives: Distinguishing Among Existing Theories," Working paper, Wharton.

- Hentschel, Ludger, and S.P. Kothari. (1995). "Life Insurance or Lottery: Are Corporations Managing or Taking Risks with Derivatives?" Working paper University of Rochester.
- Hopkins, Patrick. (1996). "The Effect of Financial Statement Classification of Hybrid Financial Instruments on Financial Analysts' Stock Price Judgments," Working paper Indiana University.
- Kormendi, R. and R. Lipe (1987). "Earnings Innovations, Earnings Persistence, and Stock Returns," <u>Journal of Business</u> vol. 60, pp. 323-346.
- Korwar, A.N., (1981). "The effect of new issues of equity: An empirical examination," Working paper University of California, Los Angeles.
- Loudder, M.L., I. Khurana, and J. Boatsman (1996). "Market Valuation of Regulatory Assets in Public Utility Firms," <u>The Accounting Review</u>, vol. 71, no.3, pp. 357-373.
- Lipe, Robert (1990). "The Relation Between Stock Returns and Accounting Earnings Given Alternative Information," <u>The Accounting Review</u>, vol. 65,no. 1, pp. 49-71.
- Miller, Merton., and K. Rock (1985). "Dividend Policy under Asymmetric Information,"

 The Journal of Finance September, pp.1031-1051.
- Modigliani, F., and M.H. Miller (1958). "The Cost of Capital, Corporation Finance and The Theory of Investment," <u>American Economic Review</u>, vol. 48, pp. 261-297.
- Myers, Stewart C. (1977). "Determinants of Corporate Borrowing," <u>Journal of Financial Economics</u> vol. 5(2), pp. 147-175.
- Myers, S.C., and N. Majluf. (1984). "Corporate financing and investment decisions when firms have information that investors do not have," <u>Journal of Financial</u> <u>Economics</u> vol. 3, pp. 187-221.
- Nance, D., C. Smith Jr., and C. Smithson. (1993). "On the Determinants of Corporate Hedging," <u>Journal of Finance</u>, vol. 48, pp. 267-284.
- Riffe, S. (1995). "The Valuation of Off-Balance-Sheet Financial Instrument Disclosures in the Banking Industry," Southern Methodist University working paper.

- Salamon, G. L., and T. L. Stober (1994). "Cross-quarter differences in stock price responses to earnings announcements: Fourth-quarter and seasonality influences," <u>Contemporary Accounting Research</u> vol. 11(1), pp. 297-330.
- Schrand, C. (1996), "The Association between stock-price interest rate sensitivity and disclosures about derivative instruments," <u>The Accounting Review</u>, vol. 72, pp. 87-109.
- Shimko, David C. (1992). <u>Finance in Continuous Time: A Primer</u>. Kolb Publishing Company.
- Smith, Clifford W., and R.M. Stulz (1985). "The Determinants of Firms' Hedging Policies," <u>Journal of Financial and Quantitative Analysis</u> December 1985.
- The Wharton School/Chase (1995) "Survey of Derivatives Usage Among U.S. Non-Financial Firms" (March).
- Titman, S. (1992) "Interest rate Swaps and Corporate Financing Choices, " <u>Journal of Finance</u>, vol.47, pp. 1503-1516.
- Wall, L. (1989). "Interest rate swaps in an Agency Theoretic Model with Uncertain Interest Rates," <u>Journal of Banking and Finance</u>, May, pp. 261-270..
- White, Halbert(1980)."A Heteroskedasticity-Consistent Covariance Matrix Estimator and a Direct Test for Heteroskedasticity," <u>Econometrica</u> vol. 48, pp. 817-838.
- Venkatachalam, M. (1996). "Value-relevance of banks derivatives disclosures," <u>Journal of Accounting and Economics</u> vol. 22, pp. 327-355.

VITA

NAME: Jennifer Ann Reynolds-Moehrle

BORN: Owensboro, Kentucky

June 14, 1962

DEGREES: B.S. University of Kentucky, 1984

M.B. Indiana University, 1996

Ph.D. Indiana University, 1997